ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Birg, Laura; Voßwinkel, Jan S.

Article — Published Version Emission taxes, firm relocation, and product differentiation

European Journal of Law and Economics

Provided in Cooperation with: Springer Nature

Suggested Citation: Birg, Laura; Voßwinkel, Jan S. (2021) : Emission taxes, firm relocation, and product differentiation, European Journal of Law and Economics, ISSN 1572-9990, Springer US, New York, NY, Vol. 51, Iss. 2, pp. 297-345, https://doi.org/10.1007/s10657-021-09685-7

This Version is available at: https://hdl.handle.net/10419/286934

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



https://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Emission taxes, firm relocation, and product differentiation

Laura Birg¹ · Jan S. Voßwinkel²

Accepted: 6 January 2021 / Published online: 1 March 2021 © The Author(s) 2021

Abstract

This paper studies the effect of an emission tax on the relocation decision in a duopoly with vertical product differentiation. We establish the relationship between an exogenous product quality markup, relocation cost, and emission taxation in a twocountry-setting for three cases: (a) an environmental tax set only by one country, (b) non-cooperative environmental taxation in both countries, and (c) coordinated environmental taxation. We show that a larger product quality markup and, thus, weaker competition can serve as a substitute for environmental policy as both reduce emissions. However, weaker competition makes firm relocation more likely, which results in emission shifting instead of emission reduction. The higher the product quality markup, the more likely it is that at least one firm relocates to the foreign country. Emission taxation in the foreign country changes location decisions: If also the foreign country applies an emission tax, at least one firm stays in the home country. If both governments set taxes non-cooperatively, the low-quality firm always stays in the home country. If both countries set taxes cooperatively, both firms are more likely to stay in the home country. However, relocation of the low-quality firm is a possible outcome under cooperative taxation.

Keywords Relocation \cdot Environmental policy \cdot Vertical quality differences \cdot Emission tax

JEL Classification $H23 \cdot F18 \cdot L13 \cdot Q58$

Jan S. Voßwinkel jan.vosswinkel@hfwu.de

> Laura Birg lbirg@uni-bremen.de

¹ Faculty of Human and Health Sciences, University of Bremen, Grazer Str 2, 28359 Bremen, Germany

² Faculty of Agriculture, Economics and Management, NGU Nürtingen-Geislingen University, Neckarsteige 6-10, 72622 Nürtingen, Germany

1 Introduction

The production of goods often involves environmental damage. Typically, governments adopt environmental policy measures, such as emission limits or emission taxes, to combat the damage.

Firms may react to such measures in two ways: First, they may adjust the extent of production or the way they produce. By producing less or producing in a more environmentally-friendly manner, firms may reduce the environmental impact. Second, firms may respond to environmental policy measures by relocating to countries with less strict environmental policy to avoid environmental regulation.

Relocation of firms may change the local distribution of emissions, and thus, environmental damage. If production results in local pollution, such as air pollution, soil degradation, or noise, the relocation of firms reduces local environmental damage. However, in this case, foreign pollution increases. If production results in global pollution, the damage is independent of the geographical origin of pollution. In this case, relocation changes the extent of damage neither in the home country nor in the foreign country.

The possibility of relocation, in turn, may change the strictness of environmental policy: If relocation of firms results in a welfare loss to their home countries, e.g., due to losses in tax revenue, the threat of relocation might result in relaxed environmental regulation. This applies, in particular, to global pollutants, such as greenhouse gases, because relocation of firms does not change the damage. For local pollutants, relocation has a damage-reducing effect at the original location, but a damage-increasing effect at the new location. Therefore, compared to global pollutants, the threat of relocation in the case of local pollutants provides a weaker incentive to relax the strictness of environmental regulation in the home country. Nevertheless, it may increase the strictness of environmental regulation in a foreign country, as the foreign government may try to reduce pollution or to prevent relocation.

The possibility of relocation creates a transboundary dimension even for local environmental damages and local environmental policy measures. Foreign countries are affected by local environmental policy measures in two ways: They suffer an increase in emissions, but they benefit from the inflow of firm profits, and possibly, tax revenue. These spillovers of national environmental policy result in a coordination problem for governments.

Firms do not necessarily react symmetrically to environmental policy when relocation is possible. A crucial factor for different relocation decisions of firms may be product quality differentiation. A firm that offers a higher product quality typically faces a less elastic demand than a firm that sells products of lower quality. Therefore, the firm with high-quality products is more capable of passing through the cost of environmental regulation and may be less willing to relocate. However, at the same time, a higher profit stemming from higher product quality and less elastic demand may allow a firm to pay the fixed cost of relocation more easily. Many consumer goods, such as television sets, drill machines, or lamps, are provided at different quality levels, focusing on different groups of (potential) consumers. As the effects of an environmental policy measure may depend on the extent of quality differentiation, and thus, the competitive environment, it may be helpful to see environmental policy measures in the context of the broader institutional setting including market structure and competition policy.

We analyze the effect of environmental policy on relocation decisions in a duopoly with vertical differentiation of product quality. Production involves local environmental damage that is independent of product quality. Thus, the market is characterized by two distortions: imperfect competition and an externality caused by local pollution. We establish the relationship between an exogenous product quality markup, relocation cost, and emission taxation in a two-country-setting for three cases: An environmental tax set only by one country, non-cooperative environmental taxation is typically a long-term decision which firms are not inclined to reverse too early. Democratic governments regularly lack the possibility for long-term commitment, e.g., because of limited legislative periods. Therefore, it seems plausible that environmental policy will react to relocation decisions of firms at some point.

We show that the larger the extent of quality differentiation of products, the more likely it is that at least one firm relocates. This is, the higher the product quality markup in the market, the less environmental policy instruments are able to achieve the desired effect. At the same time, product quality differentiation weakens competition and reduces quantities and, thus, emissions. Therefore, from an environmental perspective, weaker competition partly solves the environmental problem and may therefore be regarded as a substitute for environmental policy measures as both reduce emissions. However, relocation does not reduce global damage, as emissions are not reduced, but only shifted abroad.

Since the environmental policy of one country shifts emissions to another country, it seems plausible to assume that also the foreign government takes environmental damages and (re)location decisions of firms into account. Therefore, this paper also considers the environmental policy decisions of a foreign country and analyzes the interaction of two governments. Governments compete for tax revenue and firms' profits, which has a tax rate-decreasing effect. At the same time, governments avoid environmental damage, which has a tax rate-increasing effect.

Our analysis shows that environmental taxation of the foreign country changes the optimal relocation decisions of both firms compared to unilateral environmental taxation. If both governments set environmental taxes non-cooperatively, there is no Nash equilibrium in which the firm selling the low-quality product relocates to the foreign country. If any firm relocates, it is always the firm that provides high-quality products.

If both governments set tax rates cooperatively, it is more likely that no firm relocates. Also, under cooperative taxation also equilibria exist, in which the firm selling the low-quality product relocates.

When only one firm is located in each country, the optimal emission tax is zero for the country hosting the firm that provides the high-quality product when the product quality markup is sufficiently large. In this case, the distortion from weakened competition outweighs the distortion from the externality. This holds for noncooperative and cooperative taxation. There are several examples for our case of local pollutants: Land consumption is an example of a primarily local environmental damage, however, spillovers may arise from habitat loss for migrating species. National environmental policy may create a transboundary spillover effect if a firm relocates, moves its production site abroad, and increases land consumption at the new location. Similar considerations apply to soil protection. Although soil protection may be closely linked to groundwater protection, a significant part of soil protection is purely local. With air pollutants, environmental damage often depends on the local concentration. The longer the distance from the source of emission is, the lower is the concentration of air pollutants and the lower is the damage. Therefore, air pollution can also be considered as a mainly local environmental problem.

These examples of local pollutants are related to basic components in consumer products, such as electrical insulation or metals, such as copper. These components are used in many products and do not differentiate products in the eye of consumers. But the products containing these components are provided at different quality levels. Concerning these components, the environmental characteristics are independent of the quality of the final product. This applies, e.g., for television sets, drill machines, or lamps, which are provided at different quality levels.

Our paper provides a new perspective on the link between competition policy and environmental policy. Since weaker competition may be seen as substitute for environmental policy, competition policy and environmental policy should be coordinated carefully. Strengthening competition may have an adverse effect on the environment. However, weaker competition makes relocation as a response to local environmental policy more likely and thereby results in emission shifting instead of emission reduction.

From a European perspective, our results indicate that even when the environmental damage is primarily local, the possibility of relocation within the single market creates a case for policy coordination. The Industrial Emissions Directive (2010/75/EU) aims at creating a level playing field within the EU, although not all pollutants covered by the directive have a transboundary impact. Soil protection is also included in the Industrial Emissions Directive (2010/75/EU). Our paper offers additional justifications for a European coordination of local environmental policy as it points to the transboundary dimension of local pollution through relocation.

The rest of the paper is organized as follows. In the next section, we provide an overview over the related literature. Section 3 presents the basic model and our main assumptions. In Sect. 4, we analyze the effect of a unilateral environmental tax on relocation decisions for both firms for a time-consistent government. Section 5 considers an alternative timing of the game, showing results for a committed government. Section 6 analyses non-cooperative and coordinated environmental tax setting in both countries. Section 7 discusses several assumptions. Section 8 concludes.

2 Related literature

The paper that is most closely related to our analysis is Ikefuji et al. (2016), who analyze the effect of environmental tax policy and relocation choices in a two country-setting. In their model, two firms offer a homogeneous good. Consumers are located only in one country. The government sets an environmental tax for production emissions. They find that the decreasing relocation cost, i.e., intensified globalization, does not necessarily result in a lower emission tax.

Our paper differs from Ikefuji et al. (2016) in two ways, marking our contribution to the literature. First, we include vertical product differentiation in our analysis, showing that it results in asymmetric incentives for firms to relocate. Also, the welfare consequences of relocation differ, depending on which firm relocates. Second, we also analyze environmental taxation in the foreign country, studying both non-coordinated and coordinated taxation. The foreign government has an incentive to respond to environmental damages. We highlight the role of the competitive environment and analyze cooperation among governments in internalizing the environmental externality versus environmental tax competition.

Ing and Nicolai (2019) study the effect of an emission tax when firms may relocate in a model with heterogeneous firms that produce a homogeneous good and compete under Cournot competition. They show that environmental damage is higher when a dirty firm relocates from a developed country to a developing country. In their paper, firm heterogeneity stems from different production technologies, while goods are perfect substitutes. Our paper complements their model by assuming homogeneous production technology and differentiated goods.

Anand and Giraud-Carrier (2020) analyze the effect of environmental policy in oligopolistic markets, where firms offer differentiated products. They show that environmental regulation is able to reduce pollution and at the same time increase firms' profits, consumer surplus, and overall welfare. The effect is driven by the possibility of firms to use environmental policy as a device of tacit collusion and reduce production. Our paper also considers quality differentiation and allows for quantity effects, complementing the paper by Anand and Giraud-Carrier (2020) by considering relocation.

Our paper is part of the literature on environmental policy in open economies. Markusen et al. (1993) develop a model with two regions and two firms. They show that optimal environmental policy in an open economy where firms decide on their location differs from the closed economy setting. In their setting, small changes in environmental policy may result in firm relocation and large changes in welfare. Motta and Thisse (1994) also show that strict environmental policy might result in relocation of domestic firms. Rauscher (1995), however, shows that international tax competition for environmental taxes can result in tax rates that are either too low or too high from a welfare perspective. A similar finding is presented by Hoel (1997), who shows that international competition in environmental policy may result in a stricter environmental regulation than international cooperation. Greaker (2003) also shows that the possibility of firms to relocate might result in stricter environmental regulation because governments may

benefit from lower damages. Heuson (2010) discusses the implications of market power on the optimal choice of environmental policy instruments and shows that an emission tax may be preferred over emission standards. His results indicate that the choice of an environmental policy instrument and its strictness depends on market structure.

While the analysis of relocation decisions as the result of environmental policy has received some attention in the literature, the consideration of product differentiation has only received little attention so far to the best of our knowledge (see Reinaud 2008 for some basic intuitions).

Our analysis is also related to the literature on the pollution haven hypothesis, which states that firms prefer to produce in countries with less stringent environmental standards as this allows them to produce at lower cost. As a result, governments might abstain from too strict environmental policy, resulting in a race to the bottom. This intuitive idea has been challenged by theoretical studies as well as by empirical findings (see Sturm 2003; Rauscher 2005, for surveys). Richter et al. (2021) show in a general equilibrium model that emission taxes are inefficiently low, when the pollutant is transboundary and firms may relocate. When the pollutant is local, governments set first-best emission taxes. Cole and Elliott (2005) confirm the pollution haven hypothesis for foreign direct investment (FDI) from the US to Brazil and Mexico for capital-intensive industries. Xing and Kolstad (2002) find that the laxity of environmental regulation is an important factor for FDI decisions for heavily polluting industries, such as chemicals and primary metals, but is not significant for other industries.

Holzinger and Sommerer (2011) analyze 17 environmental regulations in more than 20 countries¹ for a period of 35 years and find a clear tendency for a race to the top instead of a race to the bottom. Most environmental regulations have been strengthened within this period instead of relaxed. The authors suggest that European harmonization is a major driver for this upward tendency in environmental regulation. Borghesi et al. (2020) analyze the effect of the EU Emissions Trading System (EU-ETS) on outward FDI with a special focus on Italian firms. While firms covered by the EU-ETS do not show a general tendency to relocate on average, firms exposed to international carbon leakage did increase production within their subsidiary firms abroad. This is in line with the findings by Naegele and Zaklan (2019), who do not find a tendency for carbon leakage caused by the EU-ETS. Ranocchia and Lambertini (2018) demonstrate that environmental policy measures can be designed so that firms reduce pollution, and at the same time, no incentive for relocation arises.

The counterpart of the pollution haven hypothesis is the Porter hypothesis, stating that environmental regulation has a positive effect on firms' productivity and innovative behavior (Porter 1991).² Zárate-Marco and Vallés-Giménez (2015)

¹ The authors analyze a sample of 21 European countries plus the United States, Mexico, and Japan.

² See Zárate-Marco and Vallés-Giménez (2015) for an overview of various studies that find empirical evidence for and against the Porter hypothesis.

investigate the effect of green taxes in Spain and find that environmental taxes increase productivity.

Concerning the effect of environmental policy on foreign direct investment (FDI), Elliot & Zhou (2013) present a theoretical framework showing that more stringent environmental standards may increase capital inflows. Dong et al. (2012) show in a two-country model that the effect of FDI on environmental regulation may depend on market sizes. If market sizes of both countries are small, FDI result in more stringent emission standards in the host country. If market sizes of both countries are large, FDI will not affect emission standards of the host country.

The problem of emission shifting within the EU-ETS is analyzed by Eichner and Pethig (2019). Within the EU-ETS, reducing carbon dioxide emissions by an additional instrument such as promoting green energy shifts emissions to other countries. Thus, national environmental policy in one member state increases emissions in other member states and makes decarbonization more difficult for them. Our paper differs from theirs as we do not consider a global pollutant or an additional environmental policy instrument (the EU-ETS), but analyze a local pollutant that becomes transboundary by the relocation of firms.

Ciocirlan and Yandle (2003) empirically test a political economy model of green taxation in OECD countries. They find that environmental taxes are not set in line with welfare maximization but that political economy considerations seem to play a major role. They also consider the possible relocation of firms, assuming that governments want to prevent relocation. Our paper does not focus on the positive analysis of environmental taxation but on the normative implications of relocation behavior. Therefore, we consider welfare-maximizing governments. In our setting, relocation also involves a beneficial effect for the home country, because the environmental damage is reduced.

The relationship between environmental policy, imperfect competition, and relocation also has a macroeconomic dimension. Metcalf and Stock (2020) measure the macroeconomic effect of a (carbon) emission tax and find no negative effect of the emission tax on employment and growth. De Loecker et al. (2020) show for the US that market power and the average profit rate has increased since 1980. This increase is mainly driven by an increase in markups for the firms that already have aboveaverage market power. Their results point in the direction that the phenomenon of imperfect competition may be of increasing relevance in developed countries.

3 The model

3.1 Model setup

Consider two countries j = H, F. In country H, two firms i, i = 1, 2 sell a product with different quality levels s_i . Assume without loss of generality that firm 1 is the

firm selling the high-quality product and firm 2 is the firm selling the low-quality product, i.e. $s_1 > s_2$. In what follows, we assume an exogenous quality ranking with $s_1 = \lambda > s_2 = 1$. Note that λ also denotes the product quality markup of the high-quality product, $\lambda = \frac{s_1}{s}$.

Firms sell their product only in H; there is no product market in country F.⁴

Consumers are heterogeneous with respect to their preference for quality θ , which is distributed uniformly on the interval [0,1]. The heterogeneity in preference parameter θ may be interpreted as differences in income, taste, or frequency of usage. Each consumer buys at most one unit of the most preferred good. The utility derived from no purchase is zero, while a consumer who buys one unit of the good obtains a net utility of

$$U = \theta s_i - p_i, i = H, L.$$
(1)

The marginal consumer indifferent between purchasing the high-quality good (from firm 1) and the low-quality good (from firm 2) is given by $\theta^* = \frac{p_1 - p_2}{\lambda - 1}$, the marginal consumer indifferent between purchasing the low-quality good and not buying is given by $\theta^{**} = p_2$. Demand is given as $q_1 = 1 - \theta^*$ and $q_2 = \theta^* - \theta^{**}$, respectively. The market is not covered so that not all consumers necessarily buy a product. The higher product quality differentiation (and hence, market power), the larger is the share of consumers who do not buy a good. Therefore environmental policy and relocation decisions may result in quantity effects: If, e.g., environmental policy increases the equilibrium price p_2 , then ceteris paribus more consumers abstain from buying a product.

Assume that production results in emissions of a local pollutant proportional to the output, such as air pollution, soil degradation or noise.⁵ Consider for the baseline scenario that emissions cause damage according to the damage function $D = \frac{1}{2}q_i^2$, i.e., marginal damage is given by q_i . In the discussion section, we discuss the effect of lower and higher marginal damage.

In the baseline scenario, the government in country H may levy an environmental tax τ_H on the emissions generated in H.⁶ In Sect. 6 we will allow also the government in F to impose an environmental tax τ_F on the emissions generated by the firms producing in F.

Both firms are initially located in country H but may relocate to country F at fixed cost ϕ . Therefore, a trade-off arises between bearing higher variable cost due

³ Both the product quality markup and the quality difference between the two products, $\lambda - 1$, increase in λ . In what follows, we refer to product quality markup. Referring to the quality difference instead would not change the reasoning of our model.

⁴ This assumption allows us to focus on the environmental policy in H. The same assumption is made in Ikefuji et al. (2016).

⁵ Air pollution and noise may also have transboundary effects, especially when occurring near the border. Here, however, we assume that only local concentration causes environmental damage and that firms are not located near the border.

Also soil degradation and land use can be considered here, although they are not literally "emissions".

⁶ Since one unit of output generates one unit of emissions, a tax on emissions is equivalent to a tax on output.

305

to the environmental tax in H and paying the fixed cost of relocation ϕ when relocating to and producing in F. Production technologies exhibit constant marginal costs, which are normalized to zero.

First, we analyze a scenario of a time-consistent policy, where in the first stage firms decide whether to relocate to the foreign country or not. In the second stage, the government sets a welfare-maximizing tax rate, and in the third stage, firms compete in prices. We discuss an alternative timing, where the government commits to an environmental tax rate in Sect. 5.

3.2 Main assumptions

We consider duopolistic competition in an uncovered market, i.e., quantity effects may occur. In an uncovered market not all (potential) consumers necessarily buy one unit of a good. This assumption is plausible for many consumer goods, such as television sets, drill machines, or lamps, where it can be assumed that consumers do not buy a product, if even the price of the good with the lowest quality is greater than the willingness to pay.

We assume that products are supplied at different quality levels and that the quality levels are fixed. In our context, this assumption seems to be justified as firms' decisions on product quality are long-run decisions that are independent of environmental policy or location decisions. Product design decisions are mainly driven by expectations about consumers' preferences and the cost of quality increases. Preferences are independent of the location of production. Thus, environmental policy affects the quantity produced but not the quality of the product.

We also assume that emissions are independent of product quality, i.e., product quality does not refer to environmental characteristics of production. So, while environmental damage is a production feature, quality is a consumption feature. This assumption seems plausible for the goods mentioned above where pollution stems from components produced by the chemical industry, e.g., electrical insulation, or metals, such as copper, that are used in products of any quality level and that do not differentiate products in the eye of consumers.

We consider a local pollutant, such as air pollution, soil degradation, land use, or noise, so that a transboundary effect stems from the relocation choice alone. Relocation is environmentally relevant only for a local pollutant because in this case, relocation has an impact on the damage in the home country and abroad.

4 Time-consistent policy

In this section, we consider a scenario where the government in H follows a timeconsistent strategy. Since firm relocation is a long-term decision, the assumption seems to be plausible that the government reserves the right to react to firms' decisions at some point. Following a committed policy in the long run requires a device for credible commitment that is hard to achieve in democratic policy making, e.g., because of limited legislative periods. We solve the game by backward induction.

4.1 Price competition

In the third stage of the game, firms compete in prices. The profit of each firm depends on its own location decision as well as on the location decision of its competitor. Therefore we consider all location equilibria that are possible from an ex-ante perspective: Both firms remain in country *H* (equilibrium *HH*), the firm selling the high-quality product remains in *H*, the firm selling the low-quality product relocates to *F* (equilibrium *HF*), the firm selling the high-quality product remains in *H*, the firm selling the high-quality product relocates to *F* (equilibrium *HF*), the firm selling the high-quality product relocates to *F* (equilibrium *HF*). Let Π denote total profits and π operating profits (total profits net of relocation cost), with $\Pi_1^{Hj} = \pi_1^{Hj}$, $\Pi_2^{H} = \pi_2^{jH}$ and $\Pi_1^{Fj} = \pi_1^{Fj} - \phi$, $\Pi_2^{FF} = \pi_2^{F} - \phi$. The following payoff matrix shows profits under the four possible strategy combinations.

1,2	Н	F
Н	Π_1^{HH}, Π_2^{HH}	Π_1^{HF}, Π_2^{HF}
F	Π_1^{FH}, Π_2^{FH}	Π_1^{FF},Π_2^{FF}

Equilibrium prices, quantities, and profits for the four equilibria can be found in Appendix A.1.

If both firms remain in country H (equilibrium HH), firms' profits are

$$\Pi_1^{HH} = \left(p_1^{HH} - \tau_H \right) q_1^{HH}, \\ \Pi_2^{HH} = \left(p_2^{HH} - \tau_H \right) q_2^{HH}.$$
(2)

If firm 1 remains in country H, but firm 2 relocates to country F (equilibrium HF), firms' profits are

$$\Pi_1^{HF} = \left(p_1^{HF} - \tau_H \right) q_1^{HF}, \ \Pi_2^{HF} = p_2^{HF} q_2^{HF} - \phi.$$
(3)

If firm 1 relocates to country F, but firm 2 remains in country H (equilibrium FH), firms' profits are

$$\Pi_1^{FH} = p_1^{FH} q_1^{FH} - \phi, \ \Pi_2^{FH} = \left(p_2^{FH} - \tau_H \right) q_2^{FH}.$$
(4)

If both firms relocate to country F (equilibrium FF), firms' profits are

$$\Pi_1^{FF} = p_1^{FF} q_1^{FF} - \phi, \ \Pi_2^{FF} = p_2^{FF} q_2^{FF} - \phi.$$
(5)

In the cases HH, FH, and FF, the quantity of firm 1 exceeds the quantity of firm 2. These findings also hold for the equilibrium HF if the product quality markup is sufficiently large. Therefore, in this case, the environmental damage resulting from the firm selling the high-quality product exceeds the environmental damage of the firm selling the low-quality product. Also, tax revenue collected from the firm selling the high-quality product in H is higher than tax revenue collected from the firm selling the low-quality product.

Firm 1 charges a higher price $(p_1 > p_2)$, with the price difference $p_1 - p_2$ increasing in the product quality markup λ . Therefore, environmental damage and the optimal emission tax also depend on product quality differentiation.

4.2 Environmental policy

Consider that in the second stage, the government in country *H* taxes the emissions of each firm producing in *H*. The government sets a tax rate τ_H to maximize social welfare, given as the sum of consumer surplus, firms' profits, tax revenue less the environmental damage.⁷ The environmental tax has four effects: It increases prices, decreases the quantity produced and thereby emissions, generates tax revenues, and may motivate one firm or both firms to relocate (and thereby reduce emissions and tax revenue in the home country).

Welfare in *H* and thus the welfare-maximizing choice of τ_H depends on location decisions in the first stage of the game.

If both firms remain in country H (equilibrium HH), welfare is given as

$$W_{H}^{HH} = CS_{H}^{HH} + \Pi_{1}^{HH} + \Pi_{2}^{HH} + \tau_{H}^{HH} \left(q_{1}^{HH} + q_{2}^{HH}\right) - \frac{1}{2} \left(q_{1}^{HH} + q_{2}^{HH}\right)^{2}.$$
 (6)

The resulting welfare maximizing tax rate is

$$\tau_H^{HH} = \frac{(2\lambda + 7)\lambda}{(\lambda + 1)(8\lambda + 1)}.$$
(7)

The equilibrium tax rate τ_{H}^{HH} decreases in λ . Since both firms produce in *H*, all production emissions are located in *H*. A higher degree of product differentiation weakens price competition, increases prices and profits, lowers quantities, and thereby lowers environmental damage. Thus, product differentiation can be seen as an imperfect substitute for the emission tax. As a result, the welfare-maximizing taxrate is lower if products are more differentiated and competition is relaxed.

If firm 1 remains in country H, but firm 2 relocates to country F (equilibrium HF), welfare is given as

$$W_{H}^{HF} = CS_{H}^{HF} + \Pi_{1}^{HF} + \tau_{H}(q_{1}^{HF}) - \frac{1}{2}(q_{1}^{HF})^{2}.$$
(8)

The welfare maximizing tax rate is

$$\tau_{H}^{HF} = \begin{cases} \frac{\lambda(\lambda-1)(9\lambda-4\lambda^{2}-3)}{-3\lambda-\lambda^{2}+4\lambda^{3}+1} & \text{if } \lambda < \frac{1}{8}\sqrt{33} + \frac{9}{8} \\ 0 & \text{if } \lambda \ge \frac{1}{8}\sqrt{33} + \frac{9}{8} \end{cases}.$$
(9)

 $^{^{7}}$ We assume that the government takes into account the profits of all firms that produce in its country. Thereby, we abstract from a situation where profits of a foreign firm are repatriated to the home country and are therefore part of the welfare in *H*.

If the product quality markup is sufficiently low (high), the welfare-maximizing tax rate τ_H^{HF} increases (decreases) in λ . Given that only firm 1 produces in *H*, the production emissions of firm 2 generate no environmental damage in *H*. Therefore, the marginal damage of production is lower compared to *HH* and so is the optimal tax rate. If the product quality markup is small, the optimal tax rate is also small, as a higher tax rate would drive the firm selling the high-quality product out of the market. A larger product quality markup increases the ability of firm 1 to passthrough the tax rate because demand is less elastic. Therefore the optimal tax rate increases. If the product quality markup is sufficiently large, the quantity of firm 1 is sufficiently low so that a tax is not necessary to correct an externality. The distortion from imperfect competition (lower quantity, smaller consumer surplus) outweighs the distortion from the environmental externality, so that an emission tax cannot increase welfare.

If firm 1 relocates to country F, but firm 2 remains in country H (equilibrium FH), welfare is given as

$$W_{H}^{FH} = CS_{H}^{FH} + \Pi_{2}^{FH} + \tau_{H}(q_{2}^{FH}) - \frac{1}{2}(q_{2}^{FH})^{2}.$$
 (10)

The welfare maximizing tax rate is

$$\tau_{H}^{FH} = \begin{cases} \frac{(\lambda - 1)(4\lambda - 2\lambda^{2} - 1)}{\lambda(-9\lambda + 8\lambda^{2} + 2)} & \text{if } \lambda < \frac{1}{2}\sqrt{2} + 1\\ 0 & \text{if } \lambda \ge \frac{1}{2}\sqrt{2} + 1 \end{cases}.$$
 (11)

Again, part of the emissions are located in *F*, but now it is the firm selling the high quality product that produces abroad. As in the case *HF*, the welfare-maximizing tax rate τ_H^{FH} is hump-shaped but compared to location combination *FH*, the government has to consider that it is the firm selling the low-quality product that produces in *H*. Demand for the firm selling the low-quality product is more elastic. Therefore, the optimal tax rate is lower. Also, the product quality markup, for which the optimal tax rate is zero (the distorted competition outweighs pollution), is lower than for the location choice *HF*.

Figure 1 shows optimal tax rates in equilibria HH (black line), HF (light gray), and FH (dark gray).

If both firms relocate to country F (equilibrium FF), there is no tax base in country H.

Optimal tax rates depend on the product quality markup as well as on location decisions, with tax rates being the highest if no firm relocates.

Product differentiation weakens competition and reduces quantities (and emissions). If both firms remain in H, the welfare-maximizing tax rate decreases in λ because weakened competition reduces quantities and emissions. The relocation of at least one firm decreases emissions in H, reduces the marginal damage, and lowers, therefore, the optimal tax rate. If one firm relocates to F and the level of product differentiation is sufficiently low, a positive tax rate results in market exit of the firm that remains in H because the tax increases variable cost and price competition



Fig. 1 Tax rates τ_{H}^{HH} , τ_{H}^{HF} , and τ_{H}^{FH}

is strong. The higher the level of product differentiation is, the higher is the optimal tax rate because the taxed firm is more capable of passing through the environmental tax to consumers. If the level of product differentiation is sufficiently high, the quantity decreases to a sufficient extent so that the optimal tax rate also decreases. For a high level of product differentiation, the quantity is sufficiently low, so that the optimal tax rate is zero.

Demand for the firm selling the low-quality product is more elastic than for the firm selling the high-quality product. Therefore optimal tax rates are always lower in the *FH* equilibria than in *HF* equilibria.

4.3 Location decision

In the first stage, firms decide whether or not to relocate to F based on expected profits, given the decision of the other firm. They anticipate the time-consistent emission tax in the second stage.

For each firm, the relocation decision is characterized by a trade-off whether to incur the variable cost of the tax levied on emissions or fixed cost of relocation and produce at zero marginal cost.

Both firms stay in *H* (equilibrium *HH*) if $\Pi_1^{HH} > \Pi_1^{FH}$ and $\Pi_2^{HH} > \Pi_2^{HF}$, i.e., $\phi > \phi^{HH}$. Firm 1 remains in *H*, but firm 2 relocates to *F* (equilibrium *HF*) if $\Pi_1^{HF} > \Pi_1^{FF} = \Pi_1^{HF} > \Pi_2^{HH}$, i.e., $\phi^{HF} < \phi < \overline{\phi}^{HF}$. Firm 1 relocates to *F* and firm 2 remains in *H* if $\Pi_1^{FH} > \Pi_2^{HH} > \Pi_1^{HH}$ and $\Pi_2^{FH} > \Pi_2^{FF} > \Pi_2^{FF}$, i.e., $\phi^{FH} < \phi < \overline{\phi}^{FH}$. Both firms relocate to *F* (equilibrium *FF*) if $\Pi_1^{FF} > \Pi_1^{HF}$ and $\Pi_2^{FF} > \Pi_2^{FH}$, i.e., $\phi < \phi^{FF}$. First stage equilibrium profits and cut-off values for ϕ can be found in Appendix A.1. Figure 2 illustrates the resulting Nash equilibria depending on the product quality markup λ and the cost of relocation ϕ . We identify several combinations of λ and ϕ with unique Nash equilibria *HH* (vertical dark gray stripes), *FH* (vertical light



Fig. 2 Location equilibria, unilateral taxation

gray stripes), HF (horizontal light gray stripes), and FF (horizontal dark gray stripes). Also, there are regions with two Nash equilibria FH and HF (solid light gray) and a region with no Nash equilibrium in pure strategies (white region).

The product quality markup λ is a key driver for the relocation decision. The lower the product quality markup, the stronger is price competition between firms, resulting in lower prices, higher quantities, and lower profits. An increase in the product quality markup λ weakens price competition and changes the result of the trade-off between paying the tax (variable cost) or paying the relocation costs (fixed cost). The higher the product quality markup, the more likely it is that at least one firm relocates to country *F*. A higher product quality markup weakens competition and reduces quantities, increases prices, reduces consumer surplus, increases profits and tax revenue, and reduces damage.

If relocation $\cot \phi$ is sufficiently high, it never pays off for any firm to relocate because the advantage of the lower variable cost is more than eaten up by the high fixed cost of relocation. Therefore, *HH* always results as an equilibrium. If relocation $\cot \phi$ is (close to) zero and the product quality markup is sufficiently low, both firms produce large quantities so that it pays off for both firms to reduce variable cost at the expense of the (low) fixed cost of relocation (*FF*). As demand for the firm selling the high-quality product is less elastic than for the firm selling the low-quality product, it is more able to pass-through the emission tax to consumers. Therefore, relocating for a fixed cost and thereby reducing variable cost is more attractive for the firm selling the low-quality product than for the firm selling the high-quality product. Therefore, the firm selling the low-quality product has a stronger incentive to relocate for moderate values of ϕ (*HF*). As the relocation decision creates a positive externality on the remaining firm in H by lowering the equilibrium tax rate, relocation of one firm dampens the relocation incentive for the remaining firm. As a result, HF as well as FH may be equilibria for some combinations of the product quality markup and relocation cost. For some combinations of low levels of product quality markup and moderate relocation cost, no equilibrium exists. For low levels of product quality markup, price competition is strong. For intermediate values of relocation cost, it pays off for at least one firm to relocate. However, the tax-dampening-effect of the relocation of at least one firm counterbalances the advantage of relocating to F. Therefore, there is always the incentive to deviate from a given location combination for at least one firm, resulting in no equilibrium.

5 Committed policy

In this section, we present our results for the case of a committed policy, as in Ikefuji et al. (2016). We consider a case where the government commits to an environmental tax rate in the first stage of the game. Firms decide whether or not to relocate in the second stage and compete in prices in the last stage of the game. While Ikefuji et al. (2016) analyze the case of firms providing homogeneous goods, we consider differentiated goods. In the following subsections, we present the backward induction solution of the three stages.

5.1 Price ceompetition

The last stage of the game, where firms set prices, is identical to the case described in Sect. 4.

5.2 Location decision

In the second stage, firms decide whether or not to relocate to F based on expected profits, given the government's commitment to an emission tax and the decision of the other firm. Cut-off values for ϕ can be found in Appendix A.2.

Both firms remain in *H* if $\Pi_1^{HH} > \Pi_1^{FH}$ and $\Pi_2^{\hat{H}\hat{H}} > \Pi_2^{HF}$, i.e., $\phi > \phi^{HH}$. The firm selling the high-quality product relocates to *F* and the firm selling the low-quality product remains in *H* (equilibrium *FH*) if $\Pi_1^{FH} > \Pi_1^{HH}$ and $\Pi_2^{FH} > \Pi_2^{FF}$, i.e., $\phi^{FH} < \phi < \phi^{FH}$. Both firms relocate to *F* if $\Pi_1^{FF} > \Pi_1^{HF}$ and $\Pi_2^{FH} > \Pi_2^{FH}$, i.e., $\phi < \overline{\phi} = \phi^{FF}$. There is no equilibrium in which the firm selling the high-quality product remains in *H* and the firm producing low quality relocates to *F*. For very high relocation cost, neither firm relocates. For intermediate relocation cost, it is always the firm selling the high-quality product that relocates. Both firms face a trade-off between paying the tax (variable cost) or relocating (fixed cost). Because of the higher quantity of the firm selling the high-quality product, the critical tax rate that makes relocation for a given relocation cost more favorable is lower than the critical tax rate for the firm producing low quality. Therefore, if only one firm

relocates, it is always the firm selling the high-quality product but not the firm producing low quality. So the government cannot commit to a tax rate that results in the firm selling the high-quality product staying in H and the firm producing low quality relocating to F.

5.3 Environmental policy

Consider that in the first stage, the government in country *H* commits to a tax rate τ_H to maximize social welfare, anticipating the relocation decisions in the second stage and price competition in the last stage. Equilibrium tax rates can be found in Appendix A.2.

If both firms remain in *H* (equilibrium *HH*), the welfare-maximizing tax rate τ_H^{HH} is

$$\tau_{H}^{HH} = \frac{(2\lambda + 7)\lambda}{(\lambda + 1)(8\lambda + 1)}$$

The welfare-maximizing tax rate under a committed policy is higher than under a time-consistent policy.

If the firm selling the high-quality product relocates to *F* and the firm selling the low-quality product remains in *H* (equilibrium *FH*), the welfare-maximizing tax rate τ_{H}^{FH} is

$$\tau_{H}^{FH} = \begin{cases} \frac{(\lambda-1)(4\lambda-2\lambda^{2}-1)}{\lambda(-9\lambda+8\lambda^{2}+2)} & \text{if } \lambda < \frac{1}{2}\sqrt{2}+1\\ 0 & \text{if } \lambda \ge \frac{1}{2}\sqrt{2}+1 \end{cases}$$

Again, the welfare-maximizing tax rate under a committed policy is higher than under a time-consistent policy.

The difference to the scenario presented in Sect. 3 stems from the fact that under time-consistent policy, there may be a welfare-maximizing tax rate for the case that the firm selling the high-quality product has already decided to relocate and the firm selling the low-quality product has decided to stay. However, after the government has decided on a tax rate, this combination of firms' decisions cannot be an equilibrium.

Tax rates under the committed policy for the cases where one firm relocates are higher than under the time-consistent policy. This is, under the committed policy, the government is more able to reduce local emissions by applying an emission tax than under time-consistent policy. However, from a policy perspective, long-term commitment is hard to achieve for democratic governments with limited legislative periods.

In our paper, under a committed policy, the location equilibrium HF does not exist, while it is possible in the model of Ikefuji et al. (2016) as they do not consider product quality differences. We show that under a committed policy, the firm selling the high-quality product has a higher incentive for relocation than the firm selling the low-quality good. Given that quality differentiation is a common feature in many markets, our results complement the results by Ikefuji et al. (2016).

6 Environmental taxes in both countries

So far, we have considered the case where an environmental tax is applied only in H. The government in F was assumed to be passive. This is in line with the analysis of Ikefuji et al. (2016). While this approach allows us to analyze the interaction of environmental damage, product quality markup, and relocation cost, it is plausible to assume that also the government in F applies an environmental tax. Therefore, in this section, we take environmental policy in F into account. We still abstract from consumers in Fand assume that an output market exists only in H. With respect to the damage function, we assume the damage function parameter of the baseline case in both countries and further assume that there is no damage spillover.

6.1 Non-cooperative taxation

Assume that the governments in H and F set environmental taxes non-cooperatively. Hosting a firm results in profits and potential tax revenues on the one hand and harmful emissions on the other hand. The relocation trade-off for firms is different in this case because relocation does not necessarily imply not being taxed.

6.1.1 Price competition

In the third stage, firms compete in prices. Equilibrium prices, quantities, and profits for the four equilibria can be found in Appendix A.3.

If both firms remain in country H (equilibrium HH), firms' profits are

$$\Pi_1^{HH} = \left(p_1^{HH} - \tau_H \right) q_1^{HH}, \\ \Pi_2^{HH} = \left(p_2^{HH} - \tau_H \right) q_2^{HH}.$$
(12)

If firm 1 remains in country H, but firm 2 relocates to country F (equilibrium HF), firms' profits are

$$\Pi_1^{HF} = \left(p_1^{HF} - \tau_H \right) q_1^{HF}, \, \Pi_2^{HF} = \left(p_2^{HF} - \tau_F \right) q_2^{HF} - \phi. \tag{13}$$

If firm 1 relocates to country F, but firm 2 remains in country H (equilibrium FH), firms' profits are

$$\Pi_1^{FH} = \left(p_1^{FH} - \tau_F\right) q_1^{FH} - \phi, \ \Pi_2^{FH} = \left(p_2^{FH} - \tau_H\right) q_2^{FH}.$$
 (14)

If both firms relocate to country F (equilibrium FF), firms' profits are

$$\Pi_1^{FH} = \left(p_1^{FH} - \tau_F\right) q_1^{FH} - \phi, \ \Pi_2^{FH} = \left(p_2^{FH} - \tau_H\right) q_2^{FH}.$$
(15)

6.1.2 Environmental policy

Consider first that in the second stage governments set tax rates non-cooperatively. Equilibrium tax rates can be found in Appendix A.3.

If both firms remain in country H (equilibrium HH), the welfare maximizing tax rate is

$$\tau_H^{HH,NC} = \frac{(2\lambda + 7)\lambda}{(\lambda + 1)(8\lambda + 1)},\tag{16}$$

which is the same as in the unilateral case (there is no tax base in country F).

If firm 1 remains in country H, but firm 2 relocates to country F (equilibrium HF), welfare in country H is given as

$$W_{H}^{HF} = CS_{H}^{HF} + \Pi_{1}^{HF} + \tau_{H}(q_{1}^{HF}) - \frac{1}{2}(q_{1}^{HF})^{2}$$
(17)

and welfare in country F is given as

$$W_F^{HF} = \Pi_2^{HF} + \tau_F (q_2^{HF}) - \frac{1}{2} (q_2^{HF})^2.$$
(18)

The welfare maximizing tax rates $\tau_{H}^{HF, NC}$ and $\tau_{F}^{HF, NC}$ are

$$\tau_{H}^{HF,NC} = \begin{cases} \frac{-6\lambda^{3}+17\lambda^{2}-11\lambda+1}{\lambda+6\lambda^{2}-4} & \text{if } \lambda < 1.9267 \\ 0 & \text{if } \lambda \ge 1.9267 \end{cases}$$

$$\tau_{F}^{HF,NC} = \begin{cases} \frac{2\lambda^{2}-1}{6\lambda^{3}+\lambda^{2}-4\lambda} & \text{if } \lambda < 1.9267 \\ \frac{(2\lambda^{2}-1)(\lambda-1)}{\lambda(2\lambda-1)(6\lambda-5)} & \text{if } \lambda \ge 1.9267 \end{cases}$$
(19)

Compared to our baseline case in Sect. 3, the equilibrium tax rate in H is now higher. This is the result of the positive tax rate in F. If production of firm 2 is taxed, quantities and therefore emissions shift from country F to country H. As a result, H applies a higher equilibrium tax to correct the externality and to deter emission shifting.

If firm 1 relocates to country F, but firm 2 remains in country H (equilibrium FH), welfare in country H is given as

$$W_{H}^{FH,NC} = CS_{H}^{FH,NC} + \Pi_{2}^{FH,NC} + \tau_{H} (q_{2}^{FH,NC}) - \frac{1}{2} (q_{2}^{FH,NC})^{2}.$$
 (20)

Welfare in country F is given as

$$W_F^{FH,NC} = \Pi_1^{FH,NC} + \tau_F (q_1^{FH,NC}) - \frac{1}{2} (q_1^{FH,NC})^2.$$
(21)

The welfare maximizing tax rates $\tau_{H}^{FH, NC}$ and $\tau_{F}^{FH, NC}$ are

$$\begin{split} \tau_{H}^{FH,NC} = & \begin{cases} \frac{(2\lambda^{3} - 6\lambda^{2} + 2\lambda + 1)}{-8\lambda^{3} + 3\lambda^{2} + 2\lambda} & \text{if} \quad \lambda < 2.5257 \\ 0 & \text{if} \quad \lambda \geq 2.5257 \\ \end{cases} \\ \tau_{F}^{FH,NC} = & \begin{cases} \frac{(6\lambda^{2} - 7\lambda + 2)}{8\lambda^{2} - 3\lambda - 2} & \text{if} \quad \lambda < 2.5257 \\ \frac{(6\lambda^{3} - 10\lambda^{2} + 4\lambda)}{8\lambda^{3} - 8\lambda^{2} + 1} & \text{if} \quad \lambda \geq 2.5257 \end{cases}. \end{split}$$

Again, equilibrium tax rates in H exceed equilibrium tax rates studied in Sect. 3 (with F assumed to be passive). The intuition is similar to the case HF. Compared to a passive government in F, taxation in F shifts emissions to H, resulting in higher taxes. In addition, taxation in F dampens the relocation incentive stemming from taxation in H, as firms now anticipate that emissions will also be taxed in F.

If both firms relocate to country F (equilibrium FF), welfare in country F is given as

$$W_F^{FF,NC} = \Pi_1^{FF,NC} + \Pi_2^{FF,NC} + \tau_F (q_1^{FF,NC} + q_2^{FF,NC}) - \frac{1}{2} (q_1^{FF,NC} + q_2^{FF,NC})^2.$$
(22)

The welfare maximizing tax rate $\tau_F^{FF, NC}$ is

$$\tau_F^{FF,NC} = \frac{2\lambda(5\lambda+4)}{12\lambda^2 + 14\lambda + 1}.$$

Note that the taxation decisions of H and F are similar if both firms produce in the respective country. But since there is no output market in F, there is no consumer surplus in F that has to be taken into account by the government.

6.1.3 Location decision

In the first stage, both firms decide whether or not to relocate by comparing equilibrium profits depending on location decisions.

First stage equilibrium profits and cut-off values for ϕ can be found in Appendix A.3.

Figure 3 illustrates the resulting Nash equilibria depending on the product quality markup λ and the cost of relocation ϕ for non-cooperative taxation in both countries.

If both governments apply an environmental tax, we identify several combinations of λ and ϕ with unique Nash equilibria *HH* (vertical dark gray stripes) and *FH* (vertical light gray stripes), and a region with no Nash equilibrium in pure strategies (white region). There is no *FF*-equilibrium, no *HF*-equilibrium, and no region characterized by multiple equilibria. For low values of λ , there is an region where no equilibrium exists.

Similar to Fig. 2, there is an region of prohibitively high relocation cost so that no firm relocates (*HH*). But compared to Fig. 2, the critical relocation cost that results in the *HH*-equilibrium is much lower. As a result of F also taxing emissions, the difference in variable cost in case of relocation decreases. Therefore, for a given

product quality markup, the critical relocation cost, which prevents the relocation of any firm, is lower.

There is no *HF*-equilibrium. If a firm relocates under non-cooperative taxation of both countries, it is always the firm selling the high-quality product that relocates. For a given product quality markup, the cut-off value of relocation cost for which the firm selling the high-quality product relocates at the margin (given that the firm selling the low-quality product stays at H) is higher than the cut-off value for the firm selling the low-quality product. Because of its higher profit, the firm selling the high-quality product. Because of its higher profit, the firm selling the high-quality product. Therefore, it relocates for a higher fixed cost than the firm selling the low-quality product. Therefore, it relocates for a higher fixed cost than the firm selling the low-quality product. The firm selling the low-quality product, in contrast, never has an incentive to relocate in equilibrium. Given that one firm (the firm selling the high-quality product) has relocated to F, country F has no incentive to attract the second firm as well.

6.2 Cooperative taxation

Consider a scenario where the governments in H and F set environmental taxes cooperatively to maximize joint welfare.

In the non-cooperative equilibrium, the environmental tax shifts emissions to the other country; this creates an incentive for high tax rates. However, relocation also shifts tax revenue, which provides a counterveiling incentive for lower tax rates. Under cooperative taxation, countries are able to balance these effects by balancing both externalities of taxation (damage shifting and revenue shifting).

6.2.1 Price competition

The third stage of the game, where firms compete in prices, is identical to our description in Sect. 3.

6.2.2 Environmental policy

Consider the case where in the second stage governments set tax rates cooperatively. Equilibrium tax rates can be found in Appendix A.4.

If both firms remain in *H*, welfare and the optimal tax rate are identical to the case *HH* under non-cooperative taxation.

If firm 1 remains in country H, but firm 2 relocates to country F (equilibrium HF), global welfare is given as

$$W^{HF,C} = CS_{H}^{HF,C} + \Pi_{1}^{HF,C} + \Pi_{2}^{HF,C} + \tau_{H}(q_{1}^{HF,C}) + \tau_{F}(q_{2}^{HF,C}) - \frac{1}{2}(q_{1}^{HF,C})^{2} - \frac{1}{2}(q_{2}^{HF,C})^{2}.$$
(23)

The welfare maximizing tax rates $\tau_{H}^{HF, C}$ and $\tau_{F}^{HF, C}$ are



Fig. 3 Location equilibria, bilateral, non-cooperative taxation

$$\begin{split} \tau_{H}^{HF,C} = & \begin{cases} \frac{(\lambda-2)(1-2\lambda)}{2\lambda+1} & \text{if } \lambda < 2\\ 0 & \text{if } \lambda \geq 2 \end{cases} \\ \tau_{F}^{HF,C} = & \begin{cases} \frac{1}{\lambda(2\lambda+1)} & \text{if } \lambda < 2\\ \frac{(\lambda-1)(-2\lambda+2\lambda^2-1)}{\lambda(-11\lambda+8\lambda^2+5)} & \text{if } \lambda \geq 2 \end{cases} \end{split}$$

Under cooperative taxation, the government in H sets a tax rate of zero, if the product quality markup is sufficiently large ($\lambda \ge 2$). If the level of quality differentiation is sufficiently high and, therefore, competition is sufficiently weak, in country H, the distortion from imperfect competition (lower produced quantity, smaller consumer surplus) outweighs the distortion of damaging emissions. Therefore, the optimal tax rate in H is zero. Imperfect competition by the product quality markup is here a substitute for environmental policy.

The government in F always sets positive tax rates. The quantity of the firm providing the low-quality product is never sufficiently low to make the emission tax unnecessary.

If firm 1 relocates to country F, but firm 2 remains in country H (equilibrium FH), global welfare is given as

$$W_{H}^{FH,C} = CS_{H}^{FH,C} + \Pi_{1}^{FH,C} + \Pi_{2}^{FH,C} + \tau_{H}(q_{2}^{FH,C}) + \tau_{F}(q_{1}^{FH,C}) - \frac{1}{2}(q_{2}^{FH,C})^{2} - \frac{1}{2}(q_{1}^{FH,C})^{2}.$$
(24)

Welfare in country F is given as

$$W_F^{FH,C} = \Pi_1^{FH,C} + \tau_F(q_1^{FH,C}) - \frac{1}{2}(q_1^{FH,C})^2.$$
(25)

The welfare maximizing tax rates $\tau_{H}^{FH, C}$ and $\tau_{F}^{FH, C}$ are

$$\tau_{H}^{FH,C} = \begin{cases} \frac{1}{\lambda(2\lambda+1)} & \text{if } \lambda < 2\\ \frac{(\lambda-1)(-2\lambda+2\lambda^{2}-1)}{\lambda(-11\lambda+8\lambda^{2}+5)} & \text{if } \lambda \ge 2 \end{cases}$$

$$\tau_{F}^{FH,C} = \begin{cases} \frac{(2\lambda-1)(2-\lambda)}{2\lambda+1} & \text{if } \lambda < 2\\ 0 & \text{if } \lambda \ge 2 \end{cases}.$$
(26)

Tax rates of *H* in equilibrium *FH* are identical to tax rates of *F* in equilibrium *HF*. Tax rates of *H* in equilibrium *HF* are identical to tax rates of *F* in equilibrium *FH*. This is, if each country hosts a firm under cooperation, tax rates in both countries depend only on the relative quality position of the respective domestic firm. If the product quality markup is sufficiently large ($\lambda \ge 2$), the optimal tax rate is zero in both cases for the country that hosts the firm selling the high-quality product 1. If the product quality markup is below this threshold level, a positive tax rate is applied.

If both firms relocate to country F (equilibrium FF), global welfare is given as

$$W^{FF,C} = CS_{H}^{FF,C} + \Pi_{1}^{FF,C} + \Pi_{2}^{FF,C} + \tau_{F}(q_{1}^{FF,C} + q_{2}^{FF,C}) - \frac{1}{2}(q_{1}^{FF,C})^{2} - \frac{1}{2}(q_{2}^{FF,C})^{2}.$$
(27)

The welfare maximizing tax rate $\tau_F^{FF, C}$ is

$$\tau_F^{FF,C} = \frac{2\lambda(5\lambda+4)}{12\lambda^2 + 14\lambda + 1}.$$

Figures 4 and 5 show tax rates for unilateral, non-cooperative bilateral and coordinated bilateral taxation for equilibria in that one firm relocates (equilibria *HF* and *FH*).

In the location equilibrium HF, the tax rate in H in the case of no taxation in F is hump-shaped (see Fig. 1 and Sect. 3.2). If the product quality markup is sufficiently large, weakened competition is a substitute for the emission tax. For low levels of the product quality markup, a larger product quality markup allows for higher tax rates without forcing the firm in the home country to exit the market. If also country F taxes emissions, welfare-maximizing tax rates in H are higher than under no taxation in F because the competitive disadvantage of taxation in H is mitigated by positive tax rates in F. Tax rates in H are again hump-shaped, but optimal tax rates are now positive even in the case of no product differentiation because also the foreign country applies a tax.

If tax rates are set cooperatively, optimal tax rates in H are higher than those under no cooperation for a sufficiently large product quality markup. The reason is that cooperation allows governments to reduce the overall quantity produced without competing for tax revenues. For low levels of product differentiation, tax rates are higher under no cooperation. Here, higher taxation results in higher (domestic) welfare, given the opportunity to shift emissions to country F.

In country F, taxation is lower for low levels of product differentiation than in country H and higher for higher levels. This is a response to the decreasing tax rate in H that shifts quantities (and emissions) to F. Tax rates in F decrease in the degree of product differentiation (but at a lower rate than in H) due to the quantity effect of weakened competition. If product differentiation is sufficiently large for the tax rate in H to be zero, the optimal tax rate in F increases in the degree of product differentiation. The higher the product quality markup is, the higher is the quantity of the low-quality product and, thus, the higher are emissions. Therefore a higher tax rate is necessary to correct the externality.

Tax rates in F are higher under no cooperation than under cooperation. This mirrors the tax rates in both cases in H. If H taxes emissions less under no cooperation, F has to set higher taxes to deter the export of emissions. If H sets higher tax rates under cooperation, F responds with lower tax rates.

In the location equilibrium FH, the optimal tax rate in H is hump-shaped in the case of unilateral taxation. If both governments tax emissions non-cooperatively, the optimal tax rate in F increases in the degree of product quality differentiation. Since there is no output market in F, there is no loss in consumer surplus in F to consider. Therefore, a higher product quality markup allows for a higher tax revenue in F and thereby transforms consumer surplus in H to tax revenue in F.

The optimal tax rate in H decreases in product quality differentiation. Prices increase because of the larger quality markup and the higher tax in F. As prices increase, the quantity produced decreases, and thus, emissions and the optimal tax rate also decrease. If the degree of product quality differentiation is sufficiently large and the quantity in H sufficiently low, the optimal tax rate in H is zero. If products are homogeneous, tax rates in both countries are identical under cooperative and non-cooperative taxation. Under cooperation, tax rates in F decrease in the degree of product differentiation is sufficiently high. Tax rates in H also decrease in the degree of product differentiation, but more slowly than in F. Product differentiation decreases the quantity produced and emissions. If the degree of product differentiation is sufficiently large so that the optimal tax rate in F is zero, the optimal tax rate increases in H. In this case, the tax in H compensates for the zero tax rate in F.

Under uncoordinated taxation, the emission tax is set to reduce local emissions (by reducing the quantity produced and by shifting emissions abroad) and to compete for tax revenue, while accepting a fiscal and/or environmental externality for the other country. Under coordinated taxation, governments do not compete for tax revenue or shift emissions to the other country, but they focus on the reduction and the optimal distribution of emissions, maximizing joint tax revenue, and profits. Abstaining from externalizing fiscal effects and/or environmental damage leads to a more targeted use of the tax.

6.2.3 Location decision

In the first stage, both firms decide whether or not to relocate. First stage equilibrium profits and cut-off values for ϕ can be found in Appendix A.4.

Figure 6 illustrates the resulting location Nash equilibria depending on the product quality markup λ and the cost of relocation ϕ .

If both governments apply an environmental tax in a coordinated manner to maximize joint welfare, we identify several combinations of λ and ϕ with unique Nash equilibria *HH* (vertical dark gray stripes), *FH* (vertical light gray stripes), *HF* (horizontal light gray stripes), an region characterized by multiple equilibria (solid light gray) and a region with no Nash equilibrium in pure strategies (white region). There is no *FF* -equilibrium. In the cooperative setting, the *FF*-equilibrium mirror-inverted to the *HH*-equilibrium, but both firms pay the fixed cost of relocation. Therefore, governments avoid this equilibrium in the cooperative case.

Compared to non-cooperative taxation (please note the different scaling of the axes), the region of HH-equilibria has increased, and there is a large region in which the firm selling the low-quality product relocates to F (which is never the case under non-cooperative taxation).

The firm selling the low-quality product relocates for relatively high levels of product quality differentiation and intermediate levels of relocation cost. Compared to no taxation in F, the firm selling the low-quality product relocates only at a higher product quality markup.

Note that governments do not tax profits and that consumer surplus only results in H. Therefore, for the product quality markup being sufficiently large (and the relocation cost being sufficiently low), the firm selling the high-quality product may relocate to F. In this case, H benefits from lower pollution. F suffers from pollution, but benefits from tax revenue (the tax allows to transform consumer surplus in H to tax revenue in F).

7 Discussion

In this section, we discuss several assumptions of our model.

7.1 Global pollutant

In this paper, we consider a local pollutant. Assuming a global pollutant would mean that the damage of emissions is independent of firms' location. In this case, relocation would affect the damage neither in the home country nor in the foreign country. Therefore, governments have no incentive to shift emissions abroad. Also, emission shifting does not result in negative spillovers for the foreign government, as also the damage in the foreign country remains unaffected by location.

For a global pollutant, a different problem arises: Governments, especially those of small countries, have no incentive to reduce emissions at all, neither through providing incentives for producing less or producing in a different way nor through



Fig. 4 τ_i^{jj} for unilateral, non-cooperative bilateral and coordinated bilateral taxation for HF



Fig. 5 τ_i^{jj} for unilateral, non-cooperative bilateral and coordinated bilateral taxation for *FH*

providing incentives for relocation. Instead, governments have strong incentives for freeriding.

In our case of local pollutants, governments have an interest in reducing local emissions through both channels: reducing production or changing the way of production and relocation.

7.2 Consumption emissions

We consider production emissions. However, many products can also generate environmentally harmful effects during use. For instance, more energy-consuming products are more harmful to the environment during use than less energy-consuming products. Passenger cars also differ in pollution.⁸

Consumption emissions are independent of production location, and governments have no interest in shifting emissions abroad. Environmental policy instruments such as a consumption related emission tax do not provide a relocation incentive. But they may have an effect on product design and market access.

7.3 Marginal damage

We consider a damage function with an increasing marginal damage. Therefore, the optimal tax rate depends on the quantity produced. Assuming a constant marginal damage would result in a lower optimal emission tax (that is independent of the quantity produced), as the optimal tax rate is equal to the marginal damage. The incentive for governments to induce relocation is lower. From a firms' perspective, lower tax rates result ceteris paribus in lower relocation incentives.

So far, we have considered a damage function $D = \frac{1}{2}q_i^2$ for a baseline case. A change in marginal damage affects location equilibria, but does not change the overall pattern of equilibria; see Appendix A.5 for a lower marginal damage $(D = \frac{1}{4}q_i^2)$ and a higher marginal damage $(D = \frac{3}{4}q_i^2)$, respectively.

For all three damage functions considered, a higher value for λ increases the probability that at least one firm relocates, given the optimal emission tax set by the government in *H*. A higher product quality markup weakens competition between firms and reduces quantities. When relocation cost is sufficiently low, it pays off for at least one firm to trade fixed cost of relocation for a lower variable cost. A higher (lower) marginal damage increases (decreases) the optimal tax rate, which, in turn, increases (decreases) the incentive of firms to relocate. Governments are more (less) interested in shifting emissions abroad when the marginal damage is higher (lower).

7.4 Quality-related damage

In this paper, pollution and environmental damage is independent of product quality. This could be the case if the environmental damage stems from product components that are similar in all products irrespective of the quality level, such as electrical insulation, metals such as copper etc. But it could also be considered that product quality is related to environmental damage. If more resources are needed for the production

⁸ Other examples are noise, waste generation caused by short product life cycles etc.



Fig. 6 Location equilibria, bilateral, coordinated taxation

of high-quality products, high-quality products could be considered to cause a higher environmental damage. On the other hand, it could also be assumed that the production of low-quality products causes a more severe environmental damage, because they use low-cost production processes that involve higher emissions.

Consider that low-quality products are also produced at a low environmental quality, i.e., cause a higher damage than high-quality products. If the firm selling the low-quality product is the "brown" firm, it generates higher damage and higher tax revenue. Due to the limited ability to pass-through the environmental tax, it has higher relocation incentives than the firm that provides high-quality products.

If the firm selling the high-quality product is the "brown" firm, it causes higher damage and higher tax revenue, but it is more able to increase prices. Therefore, it has a lower relocation incentive than the firm that provides low-quality products.

When the brown firm relocates, relocation has a greater effect on emission reduction in the home country. Therefore, the government is more interested in relocation of the brown firm. However, relocation also results in a greater loss in tax revenue. The joint effect on welfare is not obvious ex ante.

Consider, e.g., that the firm selling the high-quality product emits twice the amount of pollution as the low quality firm. Then it generates twice the tax revenue but due to the increasing marginal damage four times the damage. In this case, relocation of the brown firm increases welfare. This provides an incentive for governments to set a higher tax rate to induce the relocation of the brown firm.

Since the firm selling the high-quality product is more able to pass through the tax because demand for high-quality products is less elastic than demand for low-quality products, it is easier for governments to induce relocation of the firm that sells low-quality products. If the firm that sells low-quality products is the brown firm, governments may want to induce its relocation.

7.5 Market structure

In this paper, we assume two countries, two firms, and three scenarios (one country applies the emission tax, both countries apply the emission tax, both countries cooperate in taxation). While this setting is highly stylized, it allows us to analyze relocation incentives in a tractable manner.

From the perspective of one country, the rest of the world can be perceived as the "foreign" country. Of course, this "foreign" country is not homogeneous. For instance, import cost may differ depending on from where exactly goods are shipped – but this does not change our results in principle. Considering more than two countries that set environmental taxes results in a more complex interaction of governments – but the main strategic interaction between governments can also be shown for the two-country case.

Similarly, the duopolistic case is highly stylized. In principle, market structure is to be seen as a result of an evolutionary process of entry and exit decisions in which more than two firms may be involved. However, more complex interaction with more than two firms would distract the focus from the problem under consideration.

Considering more firms and more product quality levels would intensify competition between firms because products are closer substitutes. It would be more difficult for firms to pass through the cost of environmental regulation, especially if not all firms are obliged to pay an emission tax. This increases relocation incentives. However, stronger competition decreases profits, so it is harder for firms to pay the fixed cost of relocation. This decreases relocation incentives.

7.6 Macroeconomic perspective

Our analysis is based on a microeconomic model. However, emission taxation, the choice of production location, and weakened competition may also be analyzed in a macroeconomic perspective. The emission tax considered in our paper targets a local pollutant. Therefore, this very tax could be considered of having only a modest macroeconomic impact. In addition, the tax can be considered as being only one among many other taxes. The tax would be more macro-economically relevant if it affected a key industry. If almost ubiquitous greenhouse gases were taxed, the macroeconomic impact would be more important as such a tax affects many industrial activities.

8 Conclusion

This paper has analyzed the effect of an emission tax on the relocation decisions of firms when a duopolistic market is characterized by vertical differentiation of product quality. Especially, we have focused on two distortions, weakened competition caused by a product quality markup and an externality caused by a local pollutant. If it is only the home country applying an environmental tax in a time-consistent manner, various Nash-equilibria of location decisions exist depending on the product quality markup and relocation cost. The Nash-equilibria depend discontinuously on the cost of relocation and the product quality markup. The higher the product quality markup, the higher is the probability that at least one firm relocates to the foreign country.

If also the foreign country applies an emission tax and both governments set taxes non-cooperatively, the government in the home country cannot set a tax so that the firm selling the high-quality product relocates and the firm selling the low-quality product stays in the home country. But if both countries cooperate, this location equilibrium is possible.

Weaker competition caused by quality differentiation can be seen as a substitute for environmental policy as both reduce emissions. However, weaker competition makes firm relocation more likely, which results in emission shifting instead of emission reduction.

Our results also point at a link between competition policy and environmental policy. Strengthening competition may have an adverse effect on the environment. However, weaker competition makes relocation as a response to local environmental policy more likely and thereby results in emission shifting instead of emission reduction. Weaker competition may therefore need to be accompanied by a higher degree of policy coordination.

The possibility of relocation creates a transboundary dimension even for local environmental damages and local environmental policy measures. Therefore, international coordination of environmental policy may be needed even for local pollutants to avoid emission shifting. The Industrial Emissions Directive of the European Union may be seen as an attempt to coordinate environmental policy measures also for local pollutants within the European Union.

Appendix

A.1 Time consistent policy

Price competition

If both firms stay in the home country (equilibrium *HH*), equilibrium prices and quantities are

$$\begin{split} p_1^{HH} = & \frac{2\lambda(\lambda-1) + 3\lambda\tau_H}{4\lambda - 1}, \\ p_2^{HH} = & \frac{\lambda - 1 + \tau_H(1+2\lambda)}{4\lambda - 1} \\ q_1^{HH} = & \frac{2\lambda - \tau_H}{4\lambda - 1}, \\ q_2^{HH} = & \frac{(1-2\tau_H)\lambda}{4\lambda - 1}. \end{split}$$

Equilibrium profits are

$$\Pi_{1}^{HH} = \frac{(\lambda - 1)(\tau_{H} - 2\lambda)^{2}}{(4\lambda - 1)^{2}}, \ \Pi_{2}^{HH} = \frac{\lambda(\lambda - 1)(2\tau_{H} - 1)^{2}}{(4\lambda - 1)^{2}}.$$

If the firm selling the high-quality product remains in H, the firm selling the lowquality product relocates (equilibrium HF), equilibrium prices and quantities are

$$\begin{split} p_1^{HF} = & \frac{2\lambda(\lambda-1) + 2\lambda\tau_H}{4\lambda - 1}, \\ p_2^{HF} = & \frac{\left(\lambda + \tau_H - 1\right)}{4\lambda - 1}, \\ q_1^{HF} = & \frac{2\lambda(\lambda-1) - \tau_H(2\lambda - 1)}{4\lambda^2 - 5\lambda + 1}, \\ q_2^{HF} = & \lambda \frac{\lambda + \tau_H - 1}{4\lambda^2 - 5\lambda + 1}. \end{split}$$

Equilibrium profits are

$$\Pi_{1}^{HF} = \frac{\left(2\lambda(\lambda - 1) - (2\lambda - 1)\tau_{H}\right)^{2}}{(\lambda - 1)(4\lambda - 1)^{2}}, \\ \Pi_{2}^{HF} = \frac{\lambda\left(\lambda + \tau_{H} - 1\right)^{2}}{(\lambda - 1)(4\lambda - 1)^{2}} - \phi.$$

If the firm selling the high-quality product relocates, the firm selling the low-quality product stays in H (equilibrium FH), equilibrium prices and quantities are

$$p_{1}^{FH} = \frac{2\lambda(\lambda - 1) + \lambda\tau_{H}}{4\lambda - 1}, p_{2}^{FH} = \frac{\lambda + 2\lambda\tau_{H} - 1}{4\lambda - 1},$$
$$q_{1}^{FH} = \frac{\lambda(2(\lambda - 1) + \tau_{H})}{4\lambda^{2} - 5\lambda + 1}, q_{2}^{FH} = \frac{\lambda(\lambda - 1 - \tau_{H}(2\lambda - 1))}{4\lambda^{2} - 5\lambda + 1}.$$

Equilibrium profits are

$$\Pi_{1}^{FH} = \frac{\lambda^{2} (2(\lambda - 1) + \tau_{H})^{2}}{(\lambda - 1)(4\lambda - 1)^{2}} - \phi, \\ \Pi_{2}^{FH} = \frac{\lambda (\lambda - 1 - \tau_{H}(2\lambda - 1))^{2}}{(\lambda - 1)(4\lambda - 1)^{2}}.$$

If both firms relocate (equilibrium FF), equilibrium prices and quantities are

$$p_1^{FF} = \frac{2\lambda(\lambda - 1)}{4\lambda - 1}, p_2^{FF} = \frac{\lambda - 1}{4\lambda - 1},$$
$$q_1^{FF} = \frac{2\lambda}{4\lambda - 1}, q_2^{FF} = \frac{\lambda}{4\lambda - 1}.$$

Equilibrium profits are

$$\Pi_{1}^{FF} = \frac{4\lambda^{2}(\lambda - 1)}{(4\lambda - 1)^{2}} - \phi, \ \Pi_{2}^{FF} = \frac{\lambda(\lambda - 1)}{(4\lambda - 1)^{2}} - \phi.$$

Environmental policy

If both firms remain in country H (equilibrium HH), the resulting welfare maximizing tax rate is

$$\tau_H^{HH} = \frac{(2\lambda + 7)\lambda}{(\lambda + 1)(8\lambda + 1)}.$$

If firm 1 remains in country H, but firm 2 relocates to country F (equilibrium HF), the resulting welfare maximizing tax rate is

$$\tau_{H}^{HF} = \begin{cases} \frac{\lambda(\lambda-1)(9\lambda-4\lambda^{2}-3)}{-3\lambda-\lambda^{2}+4\lambda^{3}+1} & \text{if } \lambda < \frac{1}{8}\sqrt{33} + \frac{9}{8} \\ 0 & \text{if } \lambda \ge \frac{1}{8}\sqrt{33} + \frac{9}{8} \end{cases}$$

If firm 1 relocates to country F, but firm 2 remains in country H (equilibrium FH), the resulting welfare maximizing tax rate is

$$\tau_{H}^{FH} = \begin{cases} \frac{(\lambda - 1)(4\lambda - 2\lambda^{2} - 1)}{\lambda(-9\lambda + 8\lambda^{2} + 2)} & \text{if } \lambda < \frac{1}{2}\sqrt{2} + 1\\ 0 & \text{if } \lambda \ge \frac{1}{2}\sqrt{2} + 1 \end{cases}.$$

Location decision

If both firms stay in country H (equilibrium HH), equilibrium profits are

$$\pi_1^{HH} = \frac{\lambda^2 (\lambda - 1)(4\lambda + 5)^2}{(\lambda + 1)^2 (8\lambda + 1)^2}, \, \pi_2^{HH} = \frac{\lambda (\lambda - 1)^3}{(8\lambda^2 + 9\lambda + 1)^2}.$$

If the firm selling the high-quality product stays in H and the firm selling the lowquality product relocates to F (equilibrium HF), equilibrium profits are given as

$$\pi_{1}^{HF} = \begin{cases} \frac{\lambda^{2}(\lambda-1)^{3}(4\lambda-1)^{2}}{(-4\lambda^{3}+\lambda^{2}+3\lambda-1)^{2}} & \text{if} \quad \lambda < \frac{1}{8}\sqrt{33} + \frac{9}{8} \\ \frac{4\lambda^{2}(\lambda-1)}{(4\lambda-1)^{2}} & \text{if} \quad \lambda \ge \frac{1}{8}\sqrt{33} + \frac{9}{8} \end{cases}$$
$$\pi_{2}^{HF} = \begin{cases} \frac{\lambda(\lambda-1)(2\lambda-1)^{2}}{(-4\lambda^{3}+\lambda^{2}+3\lambda-1)^{2}} & \text{if} \quad \lambda < \frac{1}{8}\sqrt{33} + \frac{9}{8} \\ \frac{\lambda(\lambda-1)}{(4\lambda-1)^{2}} & \text{if} \quad \lambda \ge \frac{1}{8}\sqrt{33} + \frac{9}{8}. \end{cases}$$

If the firm selling the high-quality product relocates to F and the firm selling the low-quality product stays in H (equilibrium FH), equilibrium profits are

$$\pi_1^{FH} = \begin{cases} \frac{(\lambda-1)(2\lambda-1)^4}{(8\lambda^2 - 9\lambda + 2)^2} & \text{if } \lambda < \frac{1}{2}\sqrt{2} + 1\\ \frac{4\lambda^2(\lambda-1)}{(4\lambda-1)^2} & \text{if } \lambda \ge \frac{1}{2}\sqrt{2} + 1 \end{cases}$$
$$\pi_2^{FH} = \begin{cases} \frac{(\lambda-1)^3(3\lambda-1)^2}{\lambda(8\lambda^2 - 9\lambda + 2)^2} & \text{if } \lambda < \frac{1}{2}\sqrt{2} + 1\\ \frac{\lambda(\lambda-1)}{(4\lambda-1)^2} & \text{if } \lambda \ge \frac{1}{2}\sqrt{2} + 1 \end{cases}$$

If both firms relocate to F (equilibrium FF), equilibrium profits are

$$\pi_1^{FF} = \frac{4\lambda^2(\lambda - 1)}{(4\lambda - 1)^2}$$
$$\pi_2^{FF} = \frac{\lambda(\lambda - 1)}{(4\lambda - 1)^2}.$$

Both firms stay in country *H* (equilibrium *HH*) if $\phi > \phi^{HH}$, with

$$\phi^{HH} = \begin{cases} \frac{\lambda(\lambda-1)(2\lambda-1)^2}{(-4\lambda^3+\lambda^2+3\lambda-1)^2} - \frac{\lambda(\lambda-1)^3}{(8\lambda^2+9\lambda+1)^2} & \text{if} \quad \lambda < 1.274\,0\\ \frac{(\lambda-1)(2\lambda-1)^4}{(8\lambda^2-9\lambda+2)^2} - \frac{\lambda^2(\lambda-1)(4\lambda+5)^2}{(\lambda+1)^2(8\lambda+1)^2} & \text{if} \quad 1.274\,0 \le \lambda < \frac{1}{2}\sqrt{2} + 1\\ \frac{4\lambda^2(\lambda-1)}{(4\lambda-1)^2} - \frac{\lambda^2(\lambda-1)(4\lambda+5)^2}{(\lambda+1)^2(8\lambda+1)^2} & \text{if} \quad \lambda \ge \frac{1}{2}\sqrt{2} + 1, \end{cases}$$

where $\phi^{HH} = \max{\{\phi_1^{HH}, \phi_2^{HH}\}}$, with

$$\begin{split} \phi_{1}^{HH} = \pi_{1}^{FH} - \pi_{1}^{HH} = \begin{cases} \frac{(\lambda - 1)(2\lambda - 1)^{4}}{(8\lambda^{2} - 9\lambda + 2)^{2}} - \frac{\lambda^{2}(\lambda - 1)(4\lambda + 5)^{2}}{(\lambda + 1)^{2}(8\lambda + 1)^{2}} & \text{if } \lambda < \frac{1}{2}\sqrt{2} + 1\\ \frac{4\lambda^{2}(\lambda - 1)}{(4\lambda - 1)^{2}} - \frac{\lambda^{2}(\lambda - 1)(4\lambda + 5)^{2}}{(\lambda + 1)^{2}(8\lambda + 1)^{2}} & \text{if } \lambda \ge \frac{1}{2}\sqrt{2} + 1 \end{cases} \\ \phi_{2}^{HH} = \pi_{2}^{HF} - \pi_{2}^{HH} = \begin{cases} \frac{\lambda(\lambda - 1)(2\lambda - 1)^{2}}{(-4\lambda^{3} + \lambda^{2} + 3\lambda - 1)^{2}} - \frac{\lambda(\lambda - 1)^{3}}{(8\lambda^{2} + 9\lambda + 1)^{2}} & \text{if } \lambda < \frac{1}{8}\sqrt{33} + \frac{9}{8} \\ \frac{\lambda(\lambda - 1)}{(4\lambda - 1)^{2}} - \frac{\lambda(\lambda - 1)^{3}}{(8\lambda^{2} + 9\lambda + 1)^{2}} & \text{if } \lambda \ge \frac{1}{8}\sqrt{33} + \frac{9}{8}. \end{cases} \end{split}$$

Firm 1 remains in *H*, but firm 2 relocates to *F* (equilibrium *HF*) if $\underline{\phi}^{HF} < \phi < \overline{\phi}^{HF}$, with

$$\begin{split} \underline{\phi}^{HF} = \pi_1^{FF} - \pi_1^{HF} = \begin{cases} \frac{4\lambda^2(\lambda-1)}{(4\lambda-1)^2} - \frac{\lambda^2(\lambda-1)^3(4\lambda-1)^2}{(-4\lambda^3+\lambda^2+3\lambda-1)^2} & \text{if } \lambda < \frac{1}{8}\sqrt{33} + \frac{9}{8} \\ 0 & \text{if } \lambda \geq \frac{1}{8}\sqrt{33} + \frac{9}{8} \end{cases} \\ \overline{\phi}^{HF} = \pi_2^{HF} - \pi_2^{HH} = \begin{cases} \frac{\lambda(\lambda-1)(2\lambda-1)^2}{(-4\lambda^3+\lambda^2+3\lambda-1)^2} - \frac{\lambda(\lambda-1)^3}{(8\lambda^2+9\lambda+1)^2} & \text{if } \lambda < \frac{1}{8}\sqrt{33} + \frac{9}{8} \\ \frac{\lambda(\lambda-1)}{(4\lambda-1)^2} - \frac{\lambda(\lambda-1)^3}{(8\lambda^2+9\lambda+1)^2} & \text{if } \lambda \geq \frac{1}{8}\sqrt{33} + \frac{9}{8}. \end{cases} \end{split}$$

Firm 1 relocates to *F* and firm 2 remains in *H* if $\underline{\phi}^{FH} < \phi < \overline{\phi}^{FH}$, with

$$\begin{split} \underline{\phi}^{FH} = \pi_2^{FF} - \pi_2^{FH} = \begin{cases} \frac{\lambda(\lambda-1)}{(4\lambda-1)^2} - \frac{(\lambda-1)^3(3\lambda-1)^2}{\lambda(8\lambda^2-9\lambda+2)^2} & \text{if } \lambda < \frac{1}{2}\sqrt{2} + 1 \\ 0 & \text{if } \lambda \ge \frac{1}{2}\sqrt{2} + 1 \end{cases} \\ \overline{\phi}^{FH} = \pi_1^{FH} - \pi_1^{HH} = \begin{cases} \frac{(\lambda-1)(2\lambda-1)^4}{(8\lambda^2-9\lambda+2)^2} - \frac{\lambda^2(\lambda-1)(4\lambda+5)^2}{(\lambda+1)^2(8\lambda+1)^2} & \text{if } \lambda < \frac{1}{2}\sqrt{2} + 1 \\ \frac{4\lambda^2(\lambda-1)}{(4\lambda-1)^2} - \frac{\lambda^2(\lambda-1)(4\lambda+5)^2}{(\lambda+1)^2(8\lambda+1)^2} & \text{if } \lambda \ge \frac{1}{2}\sqrt{2} + 1 \end{cases} \end{split}$$

Both firms relocate to *F* (equilibrium *FF*) if $\phi < \phi^{FF}$, with

$$\phi^{FF} = \begin{cases} \frac{\lambda(\lambda-1)}{(4\lambda-1)^2} - \frac{(\lambda-1)^3(3\lambda-1)^2}{\lambda(8\lambda^2 - 9\lambda + 2)^2} & \text{if} \quad \lambda < \frac{1}{2}\sqrt{2} + 1\\ 0 & \lambda \ge \frac{1}{2}\sqrt{2} + 1 \end{cases}$$

where $\phi^{FF} = \min\{\phi_1^{FF}, \phi_2^{FF}\}$, with

$$\begin{split} \phi_1^{FF} = \pi_1^{FF} - \pi_1^{HF} = \begin{cases} \frac{4\lambda^2(\lambda-1)}{(4\lambda-1)^2} - \frac{\lambda^2(\lambda-1)^3(4\lambda-1)^2}{(-4\lambda^3+\lambda^2+3\lambda-1)^2} & \text{if } \lambda < \frac{1}{8}\sqrt{33} + \frac{9}{8} \\ 0 & \text{if } \lambda \ge \frac{1}{8}\sqrt{33} + \frac{9}{8} \end{cases} \\ \phi_2^{FF} = \pi_2^{FF} - \pi_2^{FH} = \begin{cases} \frac{\lambda(\lambda-1)}{(4\lambda-1)^2} - \frac{(\lambda-1)^3(3\lambda-1)^2}{\lambda(8\lambda^2-9\lambda+2)^2} & \text{if } \lambda < \frac{1}{2}\sqrt{2} + 1 \\ 0 & \text{if } \lambda \ge \frac{1}{2}\sqrt{2} + 1. \end{cases} \end{split}$$

A.2 Committed policy

Location decision

Both firms remain in *H* if $\phi > \phi^{HH}$, with

$$\phi^{HH} = \frac{\tau_H (2\lambda - 1) \left(4\lambda(\lambda - 1) + \tau_H \right)}{(\lambda - 1)(4\lambda - 1)^2}$$

Firm 1 relocates to F and firm 2 remains in H (equilibrium FH) if $\underline{\phi}^{FH} < \phi < \overline{\phi}^{FH}$, with

$$\begin{split} \underline{\phi}^{FH} &= \pi_2^{FF} - \pi_2^{FH} = \frac{\lambda \tau_H (2\lambda - 1) \left(2(\lambda - 1) - \tau_H (2\lambda - 1) \right)}{(\lambda - 1)(4\lambda - 1)^2} \\ \overline{\phi}^{FH} &= \pi_1^{FH} - \pi_1^{HH} = \frac{\tau_H (2\lambda - 1) \left(4\lambda(\lambda - 1) + \tau_H \right)}{(\lambda - 1)(4\lambda - 1)^2} \end{split}$$

Both firms relocate if $\phi < \phi^{FF} = \frac{\lambda \tau_H(2\lambda - 1) \left(2(\lambda - 1) - \tau_H(2\lambda - 1)\right)}{(\lambda - 1)(4\lambda - 1)^2}$.

Environmental policy

If both firms remain in H (equilibrium HH), the welfare-maximizing tax rate is

$$\tau_H^{HH} = \frac{(2\lambda + 7)\lambda}{(\lambda + 1)(8\lambda + 1)}.$$

If the firm selling the high-quality product relocates to F and the firm selling the low-quality product remains in H, the welfare-maximizing tax rate is

$$\tau_{H}^{FH} = \begin{cases} \frac{(\lambda - 1)(4\lambda - 2\lambda^{2} - 1)}{\lambda(-9\lambda + 8\lambda^{2} + 2)} & \text{if } \lambda < \frac{1}{2}\sqrt{2} + 1\\ 0 & \text{if } \lambda \ge \frac{1}{2}\sqrt{2} + 1 \end{cases}$$

A.3 Environmental taxes in both countries - non-cooperative Taxation

Price competition

If both firms stay in the home country (equilibrium *HH*), equilibrium prices and quantities are

$$\begin{split} p_1^{HH} = & \frac{2\lambda(\lambda-1) + 3\lambda\tau_H}{4\lambda - 1}, \\ p_2^{HH} = & \frac{\lambda - 1 + \tau_H(1+2\lambda)}{4\lambda - 1}, \\ q_1^{HH} = & \frac{2\lambda - \tau_H}{4\lambda - 1}, \\ q_2^{HH} = & \frac{(1-2\tau_H)\lambda}{4\lambda - 1}. \end{split}$$

Equilibrium profits are

$$\Pi_{1}^{HH} = \frac{(\lambda - 1)(\tau_{H} - 2\lambda)^{2}}{(4\lambda - 1)^{2}}, \quad \Pi_{2}^{HH} = \frac{\lambda(\lambda - 1)(2\tau_{H} - 1)^{2}}{(4\lambda - 1)^{2}}.$$

Deringer

If the firm selling the high-quality product remains in H, the firm selling the lowquality product relocates (equilibrium HF), equilibrium prices and quantities are

$$\begin{split} p_1^{HF} = &\lambda \frac{2(\lambda-1) + \tau_F + 2\tau_H}{4\lambda - 1}, \\ p_2^{HF} = \frac{\lambda + \tau_H + 2\lambda\tau_F - 1}{4\lambda - 1}, \\ q_1^{HF} = &\frac{2\lambda(\lambda-1) - (2\lambda-1)\tau_H + \lambda\tau_F}{4\lambda^2 - 5\lambda + 1}, \\ q_2^{HF} = &\frac{\lambda(\lambda-1) + \lambda\tau_H - \lambda\tau_F(2\lambda-1)}{4\lambda^2 - 5\lambda + 1}. \end{split}$$

Equilibrium profits are

$$\Pi_1^{HF} = \frac{\left(2\lambda(\lambda-1) - (2\lambda-1)\tau_H + \lambda\tau_F\right)^2}{(4\lambda-1)\left(4\lambda^2 - 5\lambda + 1\right)}, \\ \Pi_2^{HF} = \frac{\lambda\left(\lambda-1 + \tau_H - (2\lambda-1)\tau_F\right)^2}{(\lambda-1)(4\lambda-1)^2} - \phi.$$

If the firm selling the high-quality product relocates, the firm selling the low-quality product stays in H (equilibrium FH), equilibrium prices and quantities are

$$p_1^{FH} = \frac{\left(2\lambda(\lambda-1)+2\lambda\tau_F+\lambda\tau_H\right)}{4\lambda-1}, p_2^{FH} = \frac{\left(\lambda-1+\tau_F+2\lambda\tau_H\right)}{4\lambda-1}$$
$$q_1^{FH} = \frac{\left(2\lambda(\lambda-1)-\tau_F(2\lambda-1)+\lambda\tau_H\right)}{4\lambda^2-5\lambda+1}, q_2^{FH} = \frac{\lambda\left(\lambda-1+\tau_F-(2\lambda-1)\tau_H\right)}{4\lambda^2-5\lambda+1}.$$

Equilibrium profits are

$$\Pi_{1}^{FH} = \frac{\left(2\lambda(\lambda-1) - \tau_{F}(2\lambda-1) + \lambda\tau_{H}\right)^{2}}{(\lambda-1)(4\lambda-1)^{2}} - \phi, \\ \Pi_{2}^{FH} = \frac{\lambda\left(\lambda-1 + \tau_{F} - (2\lambda-1)\tau_{H}\right)^{2}}{(\lambda-1)(4\lambda-1)^{2}}.$$

If both firms relocate (equilibrium FF), equilibrium prices and quantities are

$$\begin{split} p_1^{FF} = & \frac{2\lambda(\lambda-1) + 3\lambda\tau_F}{4\lambda - 1}, \\ p_2^{FF} = & \frac{\lambda - 1 + \tau_F(2\lambda + 1)}{4\lambda - 1}, \\ q_1^{FF} = & \frac{2\lambda - \tau_F}{4\lambda - 1}, \\ q_2^{FF} = & \frac{\lambda(1 - 2\tau_F)}{4\lambda - 1}. \end{split}$$

Equilibrium profits are

$$\Pi_{1}^{FF} = \frac{(\lambda - 1)(2\lambda - \tau_{F})^{2}}{(4\lambda - 1)^{2}} - \phi, \\ \Pi_{2}^{FF} = \frac{\lambda(\lambda - 1)(1 - 2\tau_{F})^{2}}{(4\lambda - 1)^{2}} - \phi.$$

Environmental policy

If both firms remain in country H (equilibrium HH), the resulting welfare maximizing tax rate is

$$\tau_{H}^{HH,NC} = \frac{(2\lambda+7)\lambda}{(\lambda+1)(8\lambda+1)}.$$

If firm 1 remains in country H, but firm 2 relocates to country F (equilibrium HF), the resulting welfare maximizing tax rates are

$$\begin{split} \tau_{H}^{HF,NC} = \begin{cases} \frac{-6\lambda^{3}+17\lambda^{2}-11\lambda+1}{\lambda+6\lambda^{2}-4} & \text{if} \quad \lambda < 1.9267\\ 0 & \text{if} \quad \lambda \geq 1.9267 \end{cases} \\ \tau_{F}^{HF,NC} = \begin{cases} \frac{2\lambda^{2}-1}{6\lambda^{3}+\lambda^{2}-4\lambda} & \text{if} \quad \lambda < 1.9267\\ \frac{(2\lambda^{2}-1)(\lambda-1)}{\lambda(2\lambda-1)(6\lambda-5)} & \text{if} \quad \lambda \geq 1.9267 \end{cases}. \end{split}$$

If firm 1 relocates to country F, but firm 2 remains in country H (equilibrium FH), the resulting welfare maximizing tax rate are

$$\begin{split} \tau_{H}^{FH,NC} = & \begin{cases} \frac{(2\lambda^{3} - 6\lambda^{2} + 2\lambda + 1)}{-8\lambda^{3} + 3\lambda^{2} + 2\lambda} & \text{if} \quad \lambda < 2.5257 \\ 0 & \text{if} \quad \lambda \geq 2.5257 \\ \end{cases} \\ \tau_{F}^{FH,NC} = & \begin{cases} \frac{(6\lambda^{2} - 7\lambda + 2)}{8\lambda^{2} - 3\lambda - 2} & \text{if} \quad \lambda < 2.5257 \\ \frac{(6\lambda^{3} - 10\lambda^{2} + 4\lambda)}{8\lambda^{3} - 8\lambda^{2} + 1} & \text{if} \quad \lambda \geq 2.5257 \end{cases}. \end{split}$$

If both firms relocate (equilibrium FF), the resulting welfare maximizing tax rate is

$$\tau_F^{FF} = \frac{2\lambda(5\lambda+4)}{12\lambda^2 + 14\lambda + 1}$$

Location decision

If both firms remain in H (equilibrium HH), profits are

$$\pi_1^{HH} = \frac{\lambda^2 (\lambda - 1)(4\lambda + 5)^2}{(8\lambda^2 + 9\lambda + 1)^2}$$
$$\pi_2^{HH} = \frac{\lambda(\lambda - 1)^3}{(8\lambda^2 + 9\lambda + 1)^2}$$

If firm 1 remains in H and firm 2 relocates to F (equilibrium HF), profits are

$$\pi_{1}^{HF} = \begin{cases} \frac{\lambda^{2}(\lambda-1)(6\lambda-5)^{2}}{(6\lambda^{2}+\lambda-4)^{2}} & \text{if } \lambda < 1.9267\\ \frac{(\lambda-1)(6\lambda^{2}-6\lambda+1)^{2}}{(12\lambda^{2}-16\lambda+5)^{2}} & \text{if } \lambda \ge 1.9267 \end{cases}$$
$$\pi_{2}^{HF} = \begin{cases} \frac{(\lambda-1)(2\lambda-1)^{2}}{\lambda(6\lambda^{2}+\lambda-4)^{2}} & \text{if } \lambda < 1.9267\\ \frac{(\lambda-1)^{3}}{\lambda(6\lambda-5)^{2}} & \text{if } \lambda \ge 1.9267 \end{cases}$$

If firm 1 relocates to F and firm 2 remains in H (equilibrium FH), profits are

$$\pi_{1}^{FH} = \begin{cases} \frac{(\lambda - 1)(2\lambda - 1)^{4}}{(-8\lambda^{2} + 3\lambda + 2)^{2}} & \text{if} \quad \lambda < 2.5257 \\ \frac{4\lambda^{2}(\lambda - 1)^{3}}{(-4\lambda^{2} + 2\lambda + 1)^{2}} & \text{if} \quad \lambda \ge 2.5257 \\ \end{cases}$$
$$\pi_{2}^{FH} = \begin{cases} \frac{(\lambda - 1)(-3\lambda^{2} + \lambda + 1)^{2}}{\lambda(-8\lambda^{2} + 3\lambda + 2)^{2}} & \text{if} \quad \lambda < 2.5257 \\ \frac{\lambda(2\lambda^{2} - 1)^{2}(\lambda - 1)}{(8\lambda^{3} - 8\lambda^{2} + 1)^{2}} & \text{if} \quad \lambda \ge 2.5257 \end{cases}$$

If both firms relocate to F (equilibrium FF), profits are

$$\pi_1^{FF} = \frac{36\lambda^2(\lambda - 1)(\lambda + 1)^2}{(12\lambda^2 + 14\lambda + 1)^2}$$
$$\pi_2^{FF} = \frac{4\lambda^4 - 3\lambda^2 - \lambda}{(12\lambda^2 + 14\lambda + 1)^2}$$

Both firms remain in *H* if $\phi > \phi^{HH}$, with

$$\phi^{HH} = \begin{cases} \frac{(\lambda - 1)(2\lambda - 1)^2}{\lambda(6\lambda^2 + \lambda - 4)^2} - \frac{\lambda(\lambda - 1)^3}{(8\lambda^2 + 9\lambda + 1)^2} & \text{if } \lambda < 1.9267\\ \frac{(\lambda - 1)^3}{\lambda(6\lambda - 5)^2} - \frac{\lambda(\lambda - 1)^3}{(8\lambda^2 + 9\lambda + 1)^2} & \text{if } \lambda \ge 1.9267 \end{cases}$$

with $\phi^{HH} = \max{\{\phi_1^{HH}, \phi_2^{HH}\}}$, where

$$\phi_{1}^{HH} = \pi_{1}^{FH} - \pi_{1}^{HH} = \begin{cases} \frac{(\lambda - 1)(2\lambda - 1)^{4}}{(-8\lambda^{2} + 3\lambda + 2)^{2}} - \frac{\lambda^{2}(\lambda - 1)(4\lambda + 5)^{2}}{(8\lambda^{2} + 9\lambda + 1)^{2}} & \text{if } \lambda < 2.5257 \\ \frac{4\lambda^{2}(\lambda - 1)^{3}}{(-4\lambda^{2} + 2\lambda + 1)^{2}} - \frac{\lambda^{2}(\lambda - 1)(4\lambda + 5)^{2}}{(8\lambda^{2} + 9\lambda + 1)^{2}} & \text{if } \lambda \ge 2.5257 \end{cases}$$
$$\phi_{2}^{HH} = \pi_{2}^{HF} - \pi_{2}^{HH} = \begin{cases} \frac{(\lambda - 1)(2\lambda - 1)^{2}}{\lambda(6\lambda^{2} + \lambda - 4)^{2}} - \frac{\lambda(\lambda - 1)^{3}}{(8\lambda^{2} + 9\lambda + 1)^{2}} & \text{if } \lambda < 1.9267 \\ \frac{(\lambda - 1)^{3}}{\lambda(6\lambda - 5)^{2}} - \frac{\lambda(\lambda - 1)^{3}}{(8\lambda^{2} + 9\lambda + 1)^{2}} & \text{if } \lambda \ge 1.9267 \end{cases}$$

Firm 1 remains in *H* and firm 2 relocates (equilibrium *HF*) if $\underline{\phi}^{HF} < \phi < \overline{\phi}^{HF}$, with

🖄 Springer

$$\underline{\phi}^{HF} = \pi_1^{FF} - \pi_1^{HF} = \begin{cases} \frac{36\lambda^2(\lambda-1)(\lambda+1)^2}{(12\lambda^2+14\lambda+1)^2} - \frac{\lambda^2(\lambda-1)(6\lambda-5)^2}{(6\lambda^2+\lambda-4)^2} & \text{if } \lambda < 1.9267 \\ \frac{36\lambda^2(\lambda-1)(\lambda+1)^2}{(12\lambda^2+14\lambda+1)^2} - \frac{(\lambda-1)(6\lambda^2-6\lambda+1)^2}{(12\lambda^2-16\lambda+5)^2} & \text{if } \lambda \ge 1.9267 \\ \hline \overline{\phi}^{HF} = \pi_2^{HF} - \pi_2^{HH} = \begin{cases} \frac{(\lambda-1)(2\lambda-1)^2}{\lambda(6\lambda^2+\lambda-4)^2} - \frac{\lambda(\lambda-1)^3}{(8\lambda^2+9\lambda+1)^2} & \text{if } \lambda < 1.9267 \\ \frac{(\lambda-1)^3}{\lambda(6\lambda-5)^2} - \frac{\lambda(\lambda-1)^3}{(8\lambda^2+9\lambda+1)^2} & \text{if } \lambda \ge 1.9267 \end{cases}$$

Firm 1 relocates to *F* and firm 2 remains in *H* (equilibrium *FH*) if $\underline{\phi}^{FH} < \phi < \overline{\phi}^{FH}$, with

$$\underline{\phi}^{FH} = \pi_2^{FF} - \pi_2^{FH} = \begin{cases} \frac{4\lambda^4 - 3\lambda^2 - \lambda}{(12\lambda^2 + 14\lambda + 1)^2} - \frac{(\lambda - 1)(-3\lambda^2 + \lambda + 1)^2}{\lambda(-8\lambda^2 + 3\lambda + 2)^2} & \text{if } \lambda < 2.5257 \\ \frac{4\lambda^4 - 3\lambda^2 - \lambda}{(12\lambda^2 + 14\lambda + 1)^2} - \frac{\lambda(2\lambda^2 - 1)^2(\lambda - 1)}{(8\lambda^3 - 8\lambda^2 + 1)^2} & \text{if } \lambda \ge 2.5257 \\ \hline \overline{\phi}^{FH} = \pi_1^{FH} - \pi_1^{HH} = \begin{cases} \frac{(\lambda - 1)(2\lambda - 1)^4}{(-8\lambda^2 + 3\lambda + 2)^2} - \frac{\lambda^2(\lambda - 1)(4\lambda + 5)^2}{(8\lambda^2 + 9\lambda + 1)^2} & \text{if } \lambda < 2.5257 \\ \frac{4\lambda^2(\lambda - 1)^3}{(-4\lambda^2 + 2\lambda + 1)^2} - \frac{\lambda^2(\lambda - 1)(4\lambda + 5)^2}{(8\lambda^2 + 9\lambda + 1)^2} & \text{if } \lambda \ge 2.5257 \end{cases}$$

Both firms relocate if $\phi < \phi^{FF} = 0$, with $\phi^{FF} = \min\{\phi_1^{FF}, \phi_2^{FF}\}$ where

$$\phi_{1}^{FF} = \pi_{1}^{FF} - \pi_{1}^{HF} = \begin{cases} \frac{36\lambda^{2}(\lambda-1)(\lambda+1)^{2}}{(12\lambda^{2}+14\lambda+1)^{2}} - \frac{\lambda^{2}(\lambda-1)(6\lambda-5)^{2}}{(6\lambda^{2}+\lambda-4)^{2}} & \text{if } \lambda < 1.9267 \\ \frac{36\lambda^{2}(\lambda-1)(\lambda+1)^{2}}{(12\lambda^{2}+14\lambda+1)^{2}} - \frac{(\lambda-1)(6\lambda^{2}-6\lambda+1)^{2}}{(12\lambda^{2}-16\lambda+5)^{2}} & \text{if } \lambda \ge 1.9267 \\ \phi_{2}^{FF} = \pi_{2}^{FF} - \pi_{2}^{FH} = \begin{cases} \frac{4\lambda^{4}-3\lambda^{2}-\lambda}{(12\lambda^{2}+14\lambda+1)^{2}} - \frac{(\lambda-1)(-3\lambda^{2}+\lambda+1)^{2}}{\lambda(-8\lambda^{2}+3\lambda+2)^{2}} & \text{if } \lambda < 2.5257 \\ \frac{4\lambda^{4}-3\lambda^{2}-\lambda}{(12\lambda^{2}+14\lambda+1)^{2}} - \frac{\lambda(2\lambda^{2}-1)^{2}(\lambda-1)}{(8\lambda^{3}-8\lambda^{2}+1)^{2}} & \text{if } \lambda \ge 2.5257 \end{cases}$$

A.4 Environmental taxes in both countries: coordinated taxation

Environmental policy

If both firms remain in country H (equilibrium HH), the resulting welfare maximizing tax rate is

$$\tau_H^{HH,C} = \frac{(2\lambda + 7)\lambda}{(\lambda + 1)(8\lambda + 1)}.$$

If firm 1 remains in country H, but firm 2 relocates to country F (equilibrium HF), the resulting welfare maximizing tax rates are

$$\begin{split} \tau_{H}^{HF,C} = \begin{cases} \frac{(\lambda-2)(1-2\lambda)}{2\lambda+1} & \text{if } \lambda < 2\\ 0 & \text{if } \lambda \geq 2 \end{cases} \\ \tau_{F}^{HF,C} = \begin{cases} \frac{1}{\lambda(2\lambda+1)} & \text{if } \lambda < 2\\ \frac{(\lambda-1)(-2\lambda+2\lambda^2-1)}{\lambda(-11\lambda+8\lambda^2+5)} & \text{if } \lambda \geq 2 \end{cases}. \end{split}$$

If firm 1 relocates to country F, but firm 2 remains in country H (equilibrium FH), the resulting welfare maximizing tax rates are

$$\begin{split} \tau_{H}^{FH,C} = \left\{ \begin{array}{ll} \frac{1}{\lambda(2\lambda+1)} & \text{if} \quad \lambda < 2 \\ \frac{(\lambda-1)(-2\lambda+2\lambda^{2}-1)}{\lambda(-11\lambda+8\lambda^{2}+5)} & \text{if} \quad \lambda \geq 2 \end{array} \right. \\ \tau_{F}^{FH,C} = \left\{ \begin{array}{ll} \frac{(2\lambda-1)(2-\lambda)}{2\lambda+1} & \text{if} \quad \lambda < 2 \\ 0 & \text{if} \quad \lambda \geq 2 \end{array} \right. \end{split}$$

If both firms relocate (equilibrium FF), the resulting welfare maximizing tax rate is

$$\tau_F^{FF,C} = \frac{2\lambda(5\lambda+4)}{12\lambda^2 + 14\lambda + 1}.$$

Location decision

If both firms remain in H (equilibrium HH), profits are

$$\pi_1^{HH} = \frac{\lambda^2 (\lambda - 1)(4\lambda + 5)^2}{(8\lambda^2 + 9\lambda + 1)^2}$$
$$\pi_2^{HH} = \frac{\lambda (\lambda - 1)^3}{(8\lambda^2 + 9\lambda + 1)^2}$$

If firm 1 remains in H and firm 2 relocates to F (equilibrium HF), profits are

$$\pi_{1}^{HF,C} = \begin{cases} \frac{(\lambda-1)(2\lambda-1)^{2}}{(2\lambda+1)^{2}} & \text{if } \lambda < 2\\ \frac{(\lambda-1)(2\lambda-1)^{4}}{(8\lambda^{2}-11\lambda+5)^{2}} & \text{if } \lambda \geq 2 \end{cases}$$
$$\pi_{2}^{HF,C} = \begin{cases} \frac{(\lambda-1)}{\lambda(2\lambda+1)^{2}} & \text{if } \lambda < 2\\ \frac{(\lambda-1)(\lambda^{2}-\lambda+1)^{2}}{\lambda(8\lambda^{2}-11\lambda+5)^{2}} & \text{if } \lambda \geq 2 \end{cases}$$

If firm 1 relocates to F and firm 2 remains in H (equilibrium FH), profits are

$$\pi_{1}^{FH,C} = \begin{cases} \frac{(\lambda-1)(2\lambda-1)^{2}}{(2\lambda+1)^{2}} & \text{if } \lambda < 2\\ \frac{(\lambda-1)(2\lambda-1)^{4}}{(8\lambda^{2}-11\lambda+5)^{2}} & \text{if } \lambda \ge 2 \end{cases}$$
$$\pi_{2}^{FH,C} = \begin{cases} \frac{\lambda-1}{\lambda(2\lambda+1)^{2}} & \text{if } \lambda < 2\\ \frac{(\lambda-1)(\lambda^{2}-\lambda+1)^{2}}{\lambda(8\lambda^{2}-11\lambda+5)^{2}} & \text{if } \lambda \ge 2 \end{cases}$$

If both firms relocate to F (equilibrium FF), profits are

$$\pi_{1}^{FF,C} = \frac{\lambda^{2}(\lambda - 1)(4\lambda + 5)^{2}}{\left(8\lambda^{2} + 9\lambda + 1\right)^{2}}$$
$$\pi_{2}^{FF,C} = \frac{\lambda(\lambda - 1)^{3}}{\left(8\lambda^{2} + 9\lambda + 1\right)^{2}}$$

Both firms remain in *H* if $\phi > \phi^{HH}$, with

$$\phi^{HH} = \begin{cases} \frac{(\lambda - 1)}{\lambda(2\lambda + 1)^2} - \frac{\lambda(\lambda - 1)^3}{(8\lambda^2 + 9\lambda + 1)^2} & \text{if} \quad \lambda < 1.6678\\ \frac{(\lambda - 1)(2\lambda - 1)^2}{(2\lambda + 1)^2} - \frac{\lambda^2(\lambda - 1)(4\lambda + 5)^2}{(8\lambda^2 + 9\lambda + 1)^2} & \text{if} \quad 1.6678 \le \lambda < 2\\ \frac{(\lambda - 1)(2\lambda - 1)^4}{(8\lambda^2 - 11\lambda + 5)^2} - \frac{\lambda^2(\lambda - 1)(4\lambda + 5)^2}{(8\lambda^2 + 9\lambda + 1)^2} & \text{if} \quad \lambda \ge 2 \end{cases}$$

with $\phi^{HH} = \max{\{\phi_1^{HH}, \phi_2^{HH}\}}$, where

$$\phi_{1}^{HH} = \pi_{1}^{FH} - \pi_{1}^{HH} = \begin{cases} \frac{(\lambda - 1)(2\lambda - 1)^{2}}{(2\lambda + 1)^{2}} - \frac{\lambda^{2}(\lambda - 1)(4\lambda + 5)^{2}}{(8\lambda^{2} + 9\lambda + 1)^{2}} & \text{if } \lambda < 2\\ \frac{(\lambda - 1)(2\lambda - 1)^{4}}{(8\lambda^{2} - 11\lambda + 5)^{2}} - \frac{\lambda^{2}(\lambda - 1)(4\lambda + 5)^{2}}{(8\lambda^{2} + 9\lambda + 1)^{2}} & \text{if } \lambda \ge 2 \end{cases}$$
$$\phi_{2}^{HH} = \pi_{2}^{HF} - \pi_{2}^{HH} = \begin{cases} \frac{(\lambda - 1)}{\lambda(2\lambda + 1)^{2}} - \frac{\lambda(\lambda - 1)^{3}}{(8\lambda^{2} + 9\lambda + 1)^{2}} & \text{if } \lambda < 2\\ \frac{(\lambda - 1)(\lambda^{2} - \lambda + 1)^{2}}{\lambda(8\lambda^{2} - 11\lambda + 5)^{2}} - \frac{\lambda(\lambda - 1)^{3}}{(8\lambda^{2} + 9\lambda + 1)^{2}} & \text{if } \lambda \ge 2 \end{cases}$$

Firm 1 remains in *H* and firm 2 relocates (equilibrium *HF*) if $\underline{\phi}^{HF} < \phi < \overline{\phi}^{HF}$, with

$$\underline{\phi}^{HF} = \pi_1^{FF} - \pi_1^{HF} = \begin{cases} \frac{\lambda^2 (\lambda - 1)(4\lambda + 5)^2}{(8\lambda^2 + 9\lambda + 1)^2} - \frac{(\lambda - 1)(2\lambda - 1)^2}{(2\lambda + 1)^2} & \text{if } \lambda < 2\\ \frac{\lambda^2 (\lambda - 1)(4\lambda + 5)^2}{(8\lambda^2 + 9\lambda + 1)^2} - \frac{(\lambda - 1)(2\lambda - 1)^4}{(8\lambda^2 - 11\lambda + 5)^2} & \text{if } \lambda \ge 2 \end{cases}$$
$$\overline{\phi}^{HF} = \pi_2^{HF} - \pi_2^{HH} = \begin{cases} \frac{(\lambda - 1)}{\lambda(2\lambda + 1)^2} - \frac{\lambda(\lambda - 1)^3}{(8\lambda^2 + 9\lambda + 1)^2} & \text{if } \lambda < 2\\ \frac{(\lambda - 1)(\lambda^2 - \lambda + 1)^2}{\lambda(8\lambda^2 - 11\lambda + 5)^2} - \frac{\lambda(\lambda - 1)^3}{(8\lambda^2 + 9\lambda + 1)^2} & \text{if } \lambda \ge 2 \end{cases}$$

Firm 1 relocates to *F* and firm 2 remains in *H* (equilibrium *FH*) if $\underline{\phi}^{FH} < \phi < \overline{\phi}^{FH}$, with

$$\underline{\phi}^{FH} = \pi_2^{FF} - \pi_2^{FH} = \begin{cases} \frac{\lambda(\lambda-1)^3}{(8\lambda^2 + 9\lambda + 1)^2} - \frac{\lambda-1}{\lambda(2\lambda+1)^2} & \text{if } \lambda < 2\\ \frac{\lambda(\lambda-1)^3}{(8\lambda^2 + 9\lambda + 1)^2} - \frac{(\lambda-1)(\lambda^2 - \lambda+1)^2}{\lambda(8\lambda^2 - 11\lambda + 5)^2} & \text{if } \lambda \ge 2 \end{cases}$$
$$\overline{\phi}^{FH} = \pi_1^{FH} - \pi_1^{HH} = \begin{cases} \frac{(\lambda-1)(2\lambda-1)^2}{(2\lambda+1)^2} - \frac{\lambda^2(\lambda-1)(4\lambda+5)^2}{(8\lambda^2 + 9\lambda + 1)^2} & \text{if } \lambda < 2\\ \frac{(\lambda-1)(2\lambda-1)^4}{(8\lambda^2 - 11\lambda + 5)^2} - \frac{\lambda^2(\lambda-1)(4\lambda+5)^2}{(8\lambda^2 + 9\lambda + 1)^2} & \text{if } \lambda \ge 2 \end{cases}$$

Both firms relocate if $\phi < \phi^{FF} = 0$, with $\phi^{FF} = \min\{\phi_1^{FF}, \phi_2^{FF}\}$ where

$$\phi_1^{FF} = \pi_1^{FF} - \pi_1^{HF} = \begin{cases} \frac{\lambda^2 (\lambda - 1)(4\lambda + 5)^2}{(8\lambda^2 + 9\lambda + 1)^2} - \frac{(\lambda - 1)(2\lambda - 1)^2}{(2\lambda + 1)^2} & \text{if } \lambda < 2\\ \frac{\lambda^2 (\lambda - 1)(4\lambda + 5)^2}{(8\lambda^2 + 9\lambda + 1)^2} - \frac{(\lambda - 1)(2\lambda - 1)^4}{(8\lambda^2 - 11\lambda + 5)^2} & \text{if } \lambda \ge 2 \end{cases}$$
$$\phi_2^{FF} = \pi_2^{FF} - \pi_2^{FH} = \begin{cases} \frac{\lambda(\lambda - 1)^3}{(8\lambda^2 + 9\lambda + 1)^2} - \frac{\lambda - 1}{\lambda(2\lambda + 1)^2} & \text{if } \lambda < 2\\ \frac{\lambda(\lambda - 1)^3}{(8\lambda^2 + 9\lambda + 1)^2} - \frac{(\lambda - 1)(\lambda^2 - \lambda + 1)^2}{\lambda(8\lambda^2 - 11\lambda + 5)^2} & \text{if } \lambda \ge 2 \end{cases}$$

A.5 Damage function

Damage function $D = \frac{1}{4}q^2$

Environmental Policy If both firms remain in country H (equilibrium HH), the resulting welfare maximizing tax rate is

$$\tau_{H}^{HH} = \begin{cases} \frac{\lambda(11-2\lambda)}{12\lambda^{2}+14\lambda+1} & \text{if } \lambda < \frac{11}{2} \\ 0 & \text{if } \lambda \geq \frac{11}{2}. \end{cases}$$

If firm 1 remains in country H, but firm 2 relocates to country F (equilibrium HF), the resulting welfare maximizing tax rate is

$$\tau_{H}^{HF} = \begin{cases} \frac{2\lambda(1-\lambda)(4\lambda^2 - 7\lambda + 2)}{8\lambda^3 - 6\lambda^2 - 2\lambda + 1} & \text{if} \quad \lambda < \frac{1}{8}\sqrt{17} + \frac{7}{8}\\ 0 & \text{if} \quad \lambda \ge \frac{1}{8}\sqrt{17} + \frac{7}{8} \end{cases}$$

If firm 1 relocates to country F, but firm 2 remains in country H (equilibrium FH), the resulting welfare maximizing tax rate is

$$\tau_{H}^{FH} = \begin{cases} \frac{(1-\lambda)(6\lambda^{2}-9\lambda+2)}{\lambda(12\lambda^{2}-14\lambda+3)} & \text{if} \quad \lambda < \frac{1}{12}\sqrt{3}\sqrt{11} + \frac{3}{4} \\ 0 & \text{if} \quad \lambda \ge \frac{1}{12}\sqrt{3}\sqrt{11} + \frac{3}{4}. \end{cases}$$

Location Decision If both firms remain in *H* (equilibrium *HH*), equilibrium profits are

$$\pi_{1}^{HH} = \begin{cases} \frac{9\lambda^{2}(\lambda-1)(2\lambda+3)^{2}}{(12\lambda^{2}+14\lambda+1)^{2}} & \text{if } \lambda < \frac{11}{2} \\ \frac{4\lambda^{2}(\lambda-1)}{(4\lambda-1)^{2}} & \text{if } \lambda \geq \frac{11}{2} \end{cases}$$
$$\pi_{2}^{HH} = \begin{cases} \frac{\lambda(\lambda-1)(4\lambda-1)^{2}}{(12\lambda^{2}+14\lambda+1)^{2}} & \text{if } \lambda < \frac{11}{2} \\ \frac{\lambda(\lambda-1)}{(4\lambda-1)^{2}} & \text{if } \lambda \geq \frac{11}{2}. \end{cases}$$

If the firm selling the high-quality product remains in H and the firm selling the lowquality product relocates to F (equilibrium HF), profits are

$$\pi_1^{HF} = \begin{cases} \frac{4\lambda^2(\lambda-1)^3(4\lambda-1)^2}{(-8\lambda^3+6\lambda^2+2\lambda-1)^2} & \text{if } \lambda < \frac{1}{8}\sqrt{17} + \frac{7}{8} \\ \frac{4\lambda^2(\lambda-1)}{(4\lambda-1)^2} & \text{if } \lambda \ge \frac{1}{8}\sqrt{17} + \frac{7}{8} \end{cases}$$
$$\pi_2^{HF} = \begin{cases} \frac{\lambda(\lambda-1)(2\lambda-1)^2}{(-8\lambda^3+6\lambda^2+2\lambda-1)^2} & \text{if } \lambda < \frac{1}{8}\sqrt{17} + \frac{7}{8} \\ \frac{\lambda(\lambda-1)}{(4\lambda-1)^2} & \text{if } \lambda \ge \frac{1}{8}\sqrt{17} + \frac{7}{8}. \end{cases}$$

If the firm selling the high-quality product relocates to F and the firm selling the lowquality product remains in H (equilibrium FH), equilibrium profits are

$$\pi_1^{FH} = \begin{cases} \frac{(\lambda - 1)(6\lambda^2 - 7\lambda + 2)^2}{(12\lambda^2 - 14\lambda + 3)^2} & \text{if } \lambda < \frac{1}{12}\sqrt{3}\sqrt{11} + \frac{3}{4} \\ \frac{4\lambda^2(\lambda - 1)}{(4\lambda - 1)^2} & \text{if } \lambda \ge \frac{1}{12}\sqrt{3}\sqrt{11} + \frac{3}{4} \\ \end{cases}$$
$$\pi_2^{FH} = \begin{cases} \frac{4(\lambda - 1)^3(3\lambda - 1)^2}{\lambda(12\lambda^2 - 14\lambda + 3)^2} & \text{if } \lambda < \frac{1}{12}\sqrt{3}\sqrt{11} + \frac{3}{4} \\ \frac{\lambda(\lambda - 1)}{(4\lambda - 1)^2} & \text{if } \lambda \ge \frac{1}{12}\sqrt{3}\sqrt{11} + \frac{3}{4}. \end{cases}$$

If both firms relocate to F (equilibrium FF), profits are

$$\pi_1^{FF} = \frac{4\lambda^2(\lambda - 1)}{(4\lambda - 1)^2}$$
$$\pi_2^{FF} = \frac{\lambda(\lambda - 1)}{(4\lambda - 1)^2}.$$

Both firms remain in *H* if $\phi > \phi^{HH}$, with

$$\phi^{HH} = \begin{cases} \frac{\lambda(\lambda-1)(2\lambda-1)^2}{(-8\lambda^3+6\lambda^2+2\lambda-1)^2} - \frac{\lambda(\lambda-1)(4\lambda-1)^2}{(12\lambda^2+14\lambda+1)^2} & \text{if } \lambda < 1.319\\ \frac{4\lambda^2(\lambda-1)}{(4\lambda-1)^2} - \frac{9\lambda^2(\lambda-1)(2\lambda+3)^2}{(12\lambda^2+14\lambda+1)^2} & \text{if } 1.319 \le \lambda < \frac{11}{2}\\ 0 & \text{if } \lambda \ge \frac{11}{2}, \end{cases}$$

with $\phi^{HH} = \max{\{\phi_1^{HH}, \phi_2^{HH}\}}$, with

$$\begin{split} \phi_{1}^{HH} &= \pi_{1}^{FH} - \pi_{1}^{HH} \\ &= \begin{cases} \frac{(\lambda - 1)(6\lambda^{2} - 7\lambda + 2)^{2}}{(12\lambda^{2} - 14\lambda + 3)^{2}} - \frac{9\lambda^{2}(\lambda - 1)(2\lambda + 3)^{2}}{(12\lambda^{2} + 14\lambda + 1)^{2}} & \text{if } \lambda < \frac{1}{12}\sqrt{3}\sqrt{11} + \frac{3}{4} \\ \frac{4\lambda^{2}(\lambda - 1)}{(4\lambda - 1)^{2}} - \frac{9\lambda^{2}(\lambda - 1)(2\lambda + 3)^{2}}{(12\lambda^{2} + 14\lambda + 1)^{2}} & \text{if } \frac{1}{12}\sqrt{3}\sqrt{11} + \frac{3}{4} \le \lambda < \frac{11}{2} \\ 0 & \text{if } \lambda \ge \frac{11}{2} \end{cases} \\ \phi_{2}^{HH} &= \pi_{2}^{HF} - \pi_{2}^{HH} \\ &= \begin{cases} \frac{\lambda(\lambda - 1)(2\lambda - 1)^{2}}{(-8\lambda^{3} + 6\lambda^{2} + 2\lambda - 1)^{2}} - \frac{\lambda(\lambda - 1)(4\lambda - 1)^{2}}{(12\lambda^{2} + 14\lambda + 1)^{2}} & \text{if } \lambda < \frac{1}{8}\sqrt{17} + \frac{7}{8} \\ \frac{\lambda(\lambda - 1)}{(4\lambda - 1)^{2}} - \frac{\lambda(\lambda - 1)(4\lambda - 1)^{2}}{(12\lambda^{2} + 14\lambda + 1)^{2}} & \text{if } \frac{1}{8}\sqrt{17} + \frac{7}{8} \le \lambda < \frac{11}{2} \\ 0 & \text{if } \frac{11}{2} \ge \lambda. \end{cases} \end{split}$$

The firm selling the high-quality product remains in *H* and the firm selling the lowquality product relocates (equilibrium *HF*) if $\underline{\phi}^{HF} < \phi < \overline{\phi}^{HF}$, with

$$\begin{split} \underline{\phi}^{HF} &= \pi_1^{FF} - \pi_1^{HF} \\ &= \begin{cases} \frac{4\lambda^2(\lambda-1)}{(4\lambda-1)^2} - \frac{4\lambda^2(\lambda-1)^3(4\lambda-1)^2}{(-8\lambda^3+6\lambda^2+2\lambda-1)^2} & \text{if } \lambda < \frac{1}{8}\sqrt{17} + \frac{7}{8} \\ 0 & \text{if } \lambda \ge \frac{1}{8}\sqrt{17} + \frac{7}{8} \end{cases} \\ \overline{\phi}^{HF} &= \pi_2^{HF} - \pi_2^{HH} \\ &= \begin{cases} \frac{\lambda(\lambda-1)(2\lambda-1)^2}{(-8\lambda^3+6\lambda^2+2\lambda-1)^2} - \frac{\lambda(\lambda-1)(4\lambda-1)^2}{(12\lambda^2+14\lambda+1)^2} & \text{if } \lambda < \frac{1}{8}\sqrt{17} + \frac{7}{8} \\ \frac{\lambda(\lambda-1)}{(4\lambda-1)^2} - \frac{\lambda(\lambda-1)(4\lambda-1)^2}{(12\lambda^2+14\lambda+1)^2} & \text{if } \frac{1}{8}\sqrt{17} + \frac{7}{8} \le \lambda < \frac{11}{2} \\ 0 & \text{if } \lambda \ge \frac{11}{2}. \end{cases} \end{split}$$

The firm selling the high-quality product relocates to *F* and the firm selling the lowquality product remains in *H* (equilibrium *FH*) if $\underline{\phi}^{FH} < \phi < \overline{\phi}^{FH}$, with

$$\begin{split} \underline{\phi}^{FH} &= \pi_2^{FF} - \pi_2^{FH} \\ &= \begin{cases} \frac{\lambda(\lambda-1)}{(4\lambda-1)^2} - \frac{4(\lambda-1)^3(3\lambda-1)^2}{\lambda(12\lambda^2 - 14\lambda+3)^2} & \text{if } \lambda < \frac{1}{12}\sqrt{3}\sqrt{11} + \frac{3}{4} \\ 0 & \text{if } \lambda \ge \frac{1}{12}\sqrt{3}\sqrt{11} + \frac{3}{4} \end{cases} \\ \overline{\phi}^{FH} &= \pi_1^{FH} - \pi_1^{HH} \\ &= \begin{cases} \frac{(\lambda-1)(6\lambda^2 - 7\lambda+2)^2}{(12\lambda^2 - 14\lambda+3)^2} - \frac{9\lambda^2(\lambda-1)(2\lambda+3)^2}{(12\lambda^2 - 14\lambda+1)^2} & \text{if } \lambda < \frac{1}{12}\sqrt{3}\sqrt{11} + \frac{3}{4} \\ \frac{4\lambda^2(\lambda-1)}{(4\lambda-1)^2} - \frac{9\lambda^2(\lambda-1)(2\lambda+3)^2}{(12\lambda^2 + 14\lambda+1)^2} & \text{if } \frac{1}{12}\sqrt{3}\sqrt{11} + \frac{3}{4} \le \lambda < \frac{11}{2} \\ 0 & \text{if } \frac{11}{2} \ge \lambda. \end{cases} \end{split}$$

Both firms relocate to *F* (equilibrium *FF*) if $\phi < \phi^{FF}$, with

$$\phi^{FF} = \begin{cases} \frac{\lambda(\lambda-1)}{(4\lambda-1)^2} - \frac{4(\lambda-1)^3(3\lambda-1)^2}{\lambda(12\lambda^2 - 14\lambda+3)^2} & \text{if} \quad \lambda < \frac{1}{12}\sqrt{3}\sqrt{11} + \frac{3}{4}\\ 0 & \text{if} \quad \lambda \ge \frac{1}{12}\sqrt{3}\sqrt{11} + \frac{3}{4}, \end{cases}$$

where $\phi^{FF} = \min\{\phi_1^{FF}, \phi_2^{FF}\}$, with

~

$$\begin{split} \phi_1^{FF} = & \pi_1^{FF} - \pi_1^{HF} = \begin{cases} \frac{4\lambda^2(\lambda-1)}{(4\lambda-1)^2} - \frac{4\lambda^2(\lambda-1)^3(4\lambda-1)^2}{(-8\lambda^3+6\lambda^2+2\lambda-1)^2} & \text{if } \lambda < \frac{1}{8}\sqrt{17} + \frac{7}{8} \\ 0 & \text{if } \lambda \geq \frac{1}{8}\sqrt{17} + \frac{7}{8} \end{cases} \\ \phi_2^{FF} = & \pi_2^{FF} - \pi_2^{FH} = \begin{cases} \frac{\lambda(\lambda-1)}{(4\lambda-1)^2} - \frac{4(\lambda-1)^3(3\lambda-1)^2}{\lambda(12\lambda^2-14\lambda+3)^2} & \text{if } \lambda < \frac{1}{12}\sqrt{3}\sqrt{11} + \frac{3}{4} \\ 0 & \text{if } \lambda \geq \frac{1}{12}\sqrt{3}\sqrt{11} + \frac{3}{4}. \end{cases} \end{split}$$

8.0.1 Damage function $D = \frac{3}{4}q^2$

Environmental Policy If both firms remain in H (equilibrium HH), the welfare-maximizing tax rate is

$$\tau_{H}^{HH} = \frac{1}{2}$$

If the firm selling the high-quality product remains in H and the firm selling the lowquality product relocates to F (equilibrium HF), the welfare-maximizing tax rate is

$$\tau_{H}^{HF} = \begin{cases} \frac{2\lambda(\lambda-1)\left(-4\lambda^{2}+11\lambda-4\right)}{(4\lambda-3)\left(2\lambda^{2}+2\lambda-1\right)} & \text{if} \quad \lambda < \frac{1}{8}\sqrt{57} + \frac{11}{8} \\ 0 & \text{if} \quad \lambda \ge \frac{1}{8}\sqrt{57} + \frac{11}{8}. \end{cases}$$

If the firm selling the high-quality product relocates to F and the firm selling the low-quality product remains in H (equilibrium FH), the welfare-maximizing tax rate is

$$\tau_{H}^{FH} = \begin{cases} \frac{(\lambda - 1)(-2\lambda^{2} + 7\lambda - 2)}{\lambda(-22\lambda + 20\lambda^{2} + 5)} & \text{if} \quad \lambda < \frac{1}{4}\sqrt{33} + \frac{7}{4} \\ 0 & \text{if} \quad \lambda \ge \frac{1}{4}\sqrt{33} + \frac{7}{4}. \end{cases}$$

Location Decision If both firms remain in H (equilibrium HH), profits are

$$\pi_1^{HH} = \frac{(\lambda - 1)}{4}$$
$$\pi_2^{HH} = 0.$$

If the firm selling the high-quality product remains in H and the firm selling the lowquality product relocates to F (equilibrium HF), profits are

$$\pi_{1}^{HF} = \begin{cases} \frac{4\lambda^{2}(\lambda-1)^{3}(4\lambda-1)^{2}}{(8\lambda^{3}+2\lambda^{2}-10\lambda+3)^{2}} & \text{if } \lambda < \frac{1}{8}\sqrt{57} + \frac{11}{8} \\ \frac{4\lambda^{2}(\lambda-1)}{(4\lambda-1)^{2}} & \text{if } \lambda \ge \frac{1}{8}\sqrt{57} + \frac{11}{8} \end{cases}$$
$$\pi_{2}^{HF} = \begin{cases} \frac{9\lambda(\lambda-1)(2\lambda-1)^{2}}{(8\lambda^{3}+2\lambda^{2}-10\lambda+3)^{2}} & \text{if } \lambda < \frac{1}{8}\sqrt{57} + \frac{11}{8} \\ \frac{\lambda(\lambda-1)}{(4\lambda-1)^{2}} & \text{if } \lambda \ge \frac{1}{8}\sqrt{57} + \frac{11}{8}. \end{cases}$$

If the firm selling the high-quality product relocates to F and the firm selling the low-quality product remains in H (equilibrium FH), profits are

$$\pi_1^{FH} = \begin{cases} \frac{(\lambda - 1)(10\lambda^2 - 9\lambda + 2)^2}{(20\lambda^2 - 22\lambda + 5)^2} & \text{if } \lambda < \frac{1}{4}\sqrt{33} + \frac{7}{4} \\ \frac{4\lambda^2(\lambda - 1)}{(4\lambda - 1)^2} & \text{if } \lambda \ge \frac{1}{4}\sqrt{33} + \frac{7}{4} \\ \frac{4(\lambda - 1)^3(3\lambda - 1)^2}{\lambda(20\lambda^2 - 22\lambda + 5)^2} & \text{if } \lambda < \frac{1}{4}\sqrt{33} + \frac{7}{4} \\ \frac{\lambda(\lambda - 1)}{(4\lambda - 1)^2} & \text{if } \lambda \ge \frac{1}{4}\sqrt{33} + \frac{7}{4}. \end{cases}$$

If both firms relocate to F (equilibrium FF), profits are

$$\pi_1^{FF} = \frac{4\lambda^2(\lambda - 1)}{(4\lambda - 1)^2}$$
$$\pi_2^{FF} = \frac{\lambda(\lambda - 1)}{(4\lambda - 1)^2}.$$

Both firms remain in *H* (equilibrium *HH*) if $\phi > \phi^{HH}$, with

$$\phi^{HH} = \begin{cases} \frac{9\lambda(\lambda-1)(2\lambda-1)^2}{(8\lambda^3+2\lambda^2-10\lambda+3)^2} & \text{if } \lambda < 1.3105\\ \frac{(\lambda-1)(10\lambda^2-9\lambda+2)^2}{(20\lambda^2-22\lambda+5)^2} - \frac{(\lambda-1)}{4} & \text{if } 1.3105 \le \lambda < \frac{1}{4}\sqrt{33} + \frac{7}{4}\\ \frac{4\lambda^2(\lambda-1)}{(4\lambda-1)^2} - \frac{(\lambda-1)}{4} & \text{if } \lambda \ge \frac{1}{4}\sqrt{33} + \frac{7}{4}, \end{cases}$$

with $\phi^{HH} = \max{\{\phi_1^{HH}, \phi_2^{HH}\}}$ where

$$\begin{split} \phi_1^{HH} &= \pi_1^{FH} - \pi_1^{HH} \\ &= \begin{cases} \frac{(\lambda - 1)(10\lambda^2 - 9\lambda + 2)^2}{(20\lambda^2 - 22\lambda + 5)^2} - \frac{(\lambda - 1)}{4} & \text{if } \lambda < \frac{1}{4}\sqrt{33} + \frac{7}{4} \\ \frac{4\lambda^2(\lambda - 1)}{(4\lambda - 1)^2} - \frac{(\lambda - 1)}{4} & \text{if } \lambda \ge \frac{1}{4}\sqrt{33} + \frac{7}{4} \end{cases} \\ \phi_2^{HH} &= \pi_2^{HF} - \pi_2^{HH} \\ &= \begin{cases} \frac{9\lambda(\lambda - 1)(2\lambda - 1)^2}{(8\lambda^3 + 2\lambda^2 - 10\lambda + 3)^2} & \text{if } \lambda < \frac{1}{8}\sqrt{57} + \frac{11}{8} \\ \frac{\lambda(\lambda - 1)}{(4\lambda - 1)^2} & \text{if } \lambda \ge \frac{1}{8}\sqrt{57} + \frac{11}{8}. \end{cases} \end{split}$$

The firm selling the high-quality product remains $\inf_{HF} H$ and the firm selling the lowquality product relocates to F(HF) if $\underline{\phi}^{HF} < \phi < \overline{\phi}^{HF}$, with

$$\begin{split} \underline{\phi}^{HF} &= \pi_1^{FF} - \pi_1^{HF} \\ &= \begin{cases} \frac{4\lambda^2(\lambda - 1)}{(4\lambda - 1)^2} - \frac{4\lambda^2(\lambda - 1)^3(4\lambda - 1)^2}{(8\lambda^3 + 2\lambda^2 - 10\lambda + 3)^2} & \text{if} \quad \lambda < \frac{1}{8}\sqrt{57} + \frac{11}{8} \\ 0 & \text{if} \quad \lambda \ge \frac{1}{8}\sqrt{57} + \frac{11}{8} \end{cases} \\ \overline{\phi}^{HF} &= \pi_2^{HF} - \pi_2^{HH} \\ &= \begin{cases} \frac{9\lambda(\lambda - 1)(2\lambda - 1)^2}{(8\lambda^3 + 2\lambda^2 - 10\lambda + 3)^2} & \text{if} \quad \lambda < \frac{1}{8}\sqrt{57} + \frac{11}{8} \\ \frac{\lambda(\lambda - 1)}{(4\lambda - 1)^2} & \text{if} \quad \lambda \ge \frac{1}{8}\sqrt{57} + \frac{11}{8}. \end{cases} \end{split}$$

The firm selling the high-quality product relocates and the firm selling the low-quality product remains in *H* if $\underline{\phi}^{FH} < \phi < \overline{\phi}^{FH}$, with

$$\begin{split} \underline{\phi}^{FH} &= \pi_2^{FF} - \pi_2^{FH} \\ &= \begin{cases} \frac{\lambda(\lambda-1)}{(4\lambda-1)^2} - \frac{4(\lambda-1)^3(3\lambda-1)^2}{\lambda(20\lambda^2-22\lambda+5)^2} & \text{if} \quad \lambda < \frac{1}{4}\sqrt{33} + \frac{7}{4} \\ 0 & \text{if} \quad \lambda \ge \frac{1}{4}\sqrt{33} + \frac{7}{4} \end{cases} \\ \overline{\phi}^{FH} &= \pi_1^{FH} - \pi_1^{HH} \\ &= \begin{cases} \frac{(\lambda-1)(10\lambda^2-9\lambda+2)^2}{(20\lambda^2-22\lambda+5)^2} - \frac{(\lambda-1)}{4} & \text{if} \quad \lambda < \frac{1}{4}\sqrt{33} + \frac{7}{4} \\ \frac{4\lambda^2(\lambda-1)}{(4\lambda-1)^2} - \frac{(\lambda-1)}{4} & \text{if} \quad \lambda \ge \frac{1}{4}\sqrt{33} + \frac{7}{4} \end{cases} \end{split}$$

Both firms relocate to *F* if $\phi < \phi^{FF}$, with

$$\phi^{FF} = \begin{cases} \frac{\lambda(\lambda-1)}{(4\lambda-1)^2} - \frac{4(\lambda-1)^3(3\lambda-1)^2}{\lambda(20\lambda^2 - 22\lambda+5)^2} & \text{if } \lambda < \frac{1}{8}\sqrt{57} + \frac{11}{8} \\ 0 & \text{if } \lambda \ge \frac{1}{8}\sqrt{57} + \frac{11}{8} \end{cases}$$

and $\phi^{FF} = \min\{\phi_1^{FF}, \phi_2^{FF}\}$, with

$$\phi_1^{FF} = \pi_1^{FF} - \pi_1^{HF} = \begin{cases} \frac{4\lambda^2(\lambda-1)}{(4\lambda-1)^2} - \frac{4\lambda^2(\lambda-1)^3(4\lambda-1)^2}{(8\lambda^3+2\lambda^2-10\lambda+3)^2} & \text{if } \lambda < \frac{1}{8}\sqrt{57} + \frac{11}{8} \\ 0 & \text{if } \lambda \ge \frac{1}{8}\sqrt{57} + \frac{11}{8} \end{cases}$$
$$\phi_2^{FF} = \pi_2^{FF} - \pi_2^{FH} = \begin{cases} \frac{\lambda(\lambda-1)}{(4\lambda-1)^2} - \frac{4(\lambda-1)^3(3\lambda-1)^2}{\lambda(20\lambda^2-22\lambda+5)^2} & \text{if } \lambda < \frac{1}{4}\sqrt{33} + \frac{7}{4} \\ 0 & \text{if } \lambda \ge \frac{1}{4}\sqrt{33} + \frac{7}{4}. \end{cases}$$

Funding Open Access funding enabled and organized by Projekt DEAL.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- Anand, K. S. & Giraud-Carrier, F. C. (2020). Pollution Regulation of Competitive Markets. *Management Science*, 66, 4193–4206.
- Borghesi, S., Franco, C., & Marin, G. (2020). Outward foreign direct investments patterns of Italian firms in the EU-ETS. Scandinavian Journal of Economics, 122, 219–256.
- Ciocirlan, C. E., & Yandle, B. (2003). The political economy of green taxation in OECD countries. European Journal of Law and Economics, 15(3), 203–218.
- Cole, M. A., & Elliott, R. J. (2005). FDI and the capital intensity of "dirty" sectors: a missing piece of the pollution haven puzzle. *Review of Development Economics*, 9(4), 530–548.
- De Loecker, J., Eeckhout, J., & Unger, G. (2020). The rise of market power and the macroeconomic implications. *The Quarterly Journal of Economics*, 135, 561–644.
- Eichner, T., & Pethig, R. (2019). EU-type carbon regulation and the waterbed effect of green energy promotion. *Energy Economics*, 80, 656–679.
- Elliott, R. J., & Zhou, Y. (2013). Environmental regulation induced foreign direct investment. Environmental and Resource Economics, 55(1), 141–158.
- Greaker, M. (2003). Strategic environmental policy when the governments are threatened by relocation. *Resource and Energy Economics*, 25(2), 141–154.
- Heuson, C. (2010). Weitzman revisited: Emission standards versus taxes with uncertain abatement costs and market power of polluting firms. *Environmental and Resource Economics*, 47(3), 349–369.

- Holzinger, K., & Sommerer, T. (2011). 'Race to the bottom' or 'Race to Brussels'? Environmental Competition in Europe. JCMS: Journal of Common Market Studies, 49(2), 315–339.
- Ing, J. & Nicolai, J. P. (2019). Dirty versus Clean Firms' Relocation under International Trade and Imperfect Competition. CER-ETH–Center of Economic Research at ETH Zurich Working Paper 19/319.
- Ikefuji, M., Itaya, J. I., & Okamura, M. (2016). Optimal emission tax with endogenous location choice of duopolistic firms. *Environmental and Resource Economics*, 65(2), 463–485.
- Markusen, J. R., Morey, E. R., & Olewiler, N. D. (1993). Environmental policy when market structure and plant locations are endogenous. *Journal of Environmental Economics and Management*, 24(1), 69–86.
- Metcalf, G. E., & Stock, J. H. (2020). Measuring the macroeconomic impact of carbon taxes. AEA Papers and Proceedings, 110, 101–06.
- Motta, M., & Thisse, J. F. (1994). Does environmental dumping lead to delocation? *European Economic Review*, 38(3), 563–576.
- Naegele, H., & Zaklan, A. (2019). Does the EU ETS cause carbon leakage in European manufacturing? Journal of Environmental Economics and Management, 93, 125–147.
- Porter, M. E. (1991). America's Green Strategy. Scientific American, 264, 168.
- Ranocchia, C. & Lambertini, L (2018). Porter Hypothesis vs Pollution Haven Hypothesis: Can there Be Environmental Policies Getting Two Eggs in One Basket?, mimeo
- Rauscher, M. (1995). Environmental regulation and the location of polluting industries. *International Tax and Public Finance*, 2(2), 229–244.
- Rauscher, M. (2005). International Trade, Foreign Investment, and the Environment. Handbook of Environmental Economics, 3(5), 1403–1456.
- Reinaud, J. (2008). Issues behind Competitiveness and Carbon Leakage Focus on Heavy Industry. IEA Information Paper, (October), 120.
- Richter, P. M., Runkel, M. & Schmidt, R. C. (2021). Strategic environmental policy and the mobility of firms, Journal of the Association of Environmental and Resource Economists, forthcoming.
- Sturm, D. M. (2003). Trade and the environment: A survey of the literature. In L. Marsiliani, M. Rauscher, & C. Withagen (Eds.), *Environmental policy in an international perspective. Economy and environment* (Vol. 26). Dordrecht: Springer.
- Xing, Y., & Kolstad, C. D. (2002). Do lax environmental regulations attract foreign investment? Environmental and Resource Economics, 21(1), 1–22.
- Zárate-Marco, A., & Vallés-Giménez, J. (2015). Environmental tax and productivity in a decentralized context: new findings on the Porter hypothesis. *European Journal of Law and Economics*, 40(2), 313–339.

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.